

Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

June 2015

# THE U.S. DEPARTMENT OF ENERGY (DOE)'S ADVANCED MANUFACTURING OFFICE PROVIDED FUNDING FOR THIS ANALYSIS AND REPORT

The DOE Office of Energy Efficiency and Renewable Energy (EERE)'s Advanced Manufacturing Office works with industry, small business, universities, and other stakeholders to identify and invest in emerging technologies with the potential to create high-quality domestic manufacturing jobs and enhance the global competitiveness of the United States.

Prepared for DOE / EERE's Advanced Manufacturing Office by Energetics Incorporated

(Keith Jamison, Caroline Kramer, Sabine Brueske, Aaron Fisher)

#### Disclaimer

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, nor any of their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof.

## **Preface**

Reducing energy consumption through investment in advanced technologies and practices can enhance American manufacturing competitiveness. Energy bandwidth studies of U.S. manufacturing sectors serve as general data references to help understand the range (or *bandwidth*) of potential energy savings opportunities. The U.S. Department of Energy (DOE)'s Advanced Manufacturing Office (AMO) has commissioned a series of bandwidth studies to analyze the processes and products that consume the most energy, and provide hypothetical, technology-based estimates of potential energy savings opportunities. The consistent methodology used in the bandwidth studies provides a framework to evaluate and compare energy savings potentials within and across manufacturing sectors at the macro-scale. Bandwidth studies using the terminology and methodology outlined below were prepared for the Chemicals, Petroleum Refining, Iron and Steel, and Pulp and Paper industry sectors in 2014.<sup>11</sup>

Four different energy bands (or measures) are used consistently in this series to describe different levels of onsite energy consumption to manufacture specific products and to compare potential energy savings opportunities in U.S. manufacturing facilities (see figure). Current typical (CT) is the energy consumption in 2010; state of the art (SOA) is the energy consumption that may be possible through the adoption of existing best technologies and practices available worldwide; practical **minimum** (PM) is the energy consumption that may be possible if applied R&D technologies under development worldwide are deployed; and the thermodynamic minimum



Energy Consumption Bands and Opportunity Bandwidths Estimated in this Study

(TM) is the least amount of energy required under ideal conditions, which typically cannot be attained in commercial applications. CT energy consumption serves as the benchmark of manufacturing energy consumption. TM energy consumption serves as the baseline (or

<sup>&</sup>lt;sup>1</sup> The concept of an energy bandwidth, and its use as an analysis tool for identifying potential energy saving opportunities, originated in AMO in 2002 (when it was called the Office of Industrial Technologies). The first two sector studies—Iron and Steel, and Metal Castings—were completed in 2004. That work was followed by Chemicals and Petroleum Refining studies in 2006, and Aluminum, Glass, and Mining in 2007. A Cement Industry analysis was conducted in 2010 and a Pulp and Paper analysis was conducted in 2011.

theoretical minimum) that is used in calculating energy savings potential. Feedstock energy (the nonfuel use of fossil energy) is not included in the energy consumption estimates.

Two onsite energy savings opportunity *bandwidths* are estimated: the *current opportunity* spans the bandwidth from CT energy consumption to SOA energy consumption, and the *R&D opportunity* spans the bandwidth from SOA energy consumption to PM energy consumption. These bandwidths are estimated for processes and products studied and for all manufacturing within a sector based on extrapolated data. The difference between PM energy consumption and TM energy consumption is labeled as *impractical*. The term *impractical* is used because with today's knowledge of technologies in R&D, further investment may no longer lead to incremental energy savings and thermodynamic limitations impede technology opportunities. Significant investment in technology development and implementation would be needed to fully realize the energy savings opportunities estimated. The costs associated with achieving SOA and PM energy consumption are not considered in this report; a techno-economic analysis of the costs and benefits of future R&D technologies was not in the scope of this study.

In each sector studied in the series, the four energy bands are estimated for select individual products or processes, subsectors, and sector-wide. The estimation method compares diverse industry, governmental, and academic data to analyses of reported plant energy consumption data from the Manufacturing Energy Consumption Survey (MECS) conducted by the U.S. Energy Information Administration (EIA). MECS is a national sample survey of U.S. manufacturing establishments conducted every four years; information is collected and reported on U.S. manufacturing energy consumption and expenditures.

## Acknowledgements

AMO wishes to acknowledge the contributions made by Arvind Thekdi of E3M and David Forrest of AMO for their work reviewing this study. Appreciation is also extended to the American Iron and Steel Institute for its valuable contributions to this report.

In addition, AMO recognizes Joseph Cresko of DOE/AMO who lead the conceptual development and publication of the bandwidth study series with the support of Dr. Alberta Carpenter at the National Renewable Energy Laboratory, as well as the important contributions made by Keith Jamison, Sabine Brueske, and Caroline Kramer of Energetics Incorporated for conducting the majority of the research and analysis and drafting this study.

## **Executive Summary**

The United States is a significant producer of iron and steel products. This bandwidth study examines energy consumption and potential energy savings opportunities in U.S. iron and steel manufacturing. Industrial, government, and academic data are used to estimate the energy consumed in six of the most energy intensive iron and steel manufacturing processes. Three different energy consumption *bands* (or levels) are estimated for these select manufacturing processes based on referenced energy intensities of current, state of the art, and R&D technologies. A fourth theoretical minimum energy consumption *band* is also estimated. The data from the select processes studied is also extrapolated to determine energy consumption for the entire iron and steel sector. The *bandwidth*—the difference between bands of energy consumption—is used to determine the potential energy savings opportunity. The costs associated with realizing these energy savings was not in the scope of this study.

The purpose of this data analysis is to provide macro-scale estimates of energy savings opportunities for iron and steel manufacturing processes and sector-wide. This is a step toward understanding the processes that could most benefit from technology and efficiency improvements to realize energy savings.

*Study Organization and Approach:* After providing an overview of the methodology (Chapter 1) and energy consumption in iron and steel manufacturing (Chapter 2), the 2010 production volumes (Chapter 3) and current energy consumption (current typical [CT], Chapter 4) were estimated for six select processes. In addition, the minimum energy consumption for these processes was estimated assuming the adoption of best technologies and practices available worldwide (state of the art [SOA], Chapter 5) and assuming the deployment of the applied research and development (R&D) technologies available worldwide (practical minimum [PM], Chapter 6). The minimum amount of energy theoretically required for these processes assuming ideal conditions was also estimated (thermodynamic minimum [TM)], Chapter 7); in some cases, this is less than zero. The difference between the energy consumption *bands* (CT, SOA, PM, TM) are the estimated energy savings opportunity *bandwidths* (Chapter 8).

The U.S. Energy Information Administration's (EIA) Manufacturing Energy Consumption Survey (MECS) provides a sector-wide estimate of energy consumption for U.S. iron and steel manufacturing; this data is referenced as sector-wide CT energy consumption. In this study, CT, SOA, PM, and TM energy consumption for six *individual* processes is estimated from multiple referenced sources. To estimate SOA, PM, and TM energy consumption for the entire sector, the CT, SOA, PM, and TM energy consumption data of the six processes studies is extrapolated estimate total sector-wide SOA, PM, and TM energy consumption. In 2010, these six processes corresponded to 82% of the industry's energy consumption. *Study Results:* Two energy savings opportunity *bandwidths* – current opportunity and R&D opportunity – are presented in Table ES-1 and Figure ES-1.<sup>1</sup> The current opportunity is the difference between the 2010 CT energy consumption and SOA energy consumption; the R&D opportunity is the difference between SOA energy consumption and PM energy consumption. Potential energy savings opportunities are presented for the six processes studied and for all of U.S. iron and steel manufacturing based on extrapolated data. Figure ES-1 also shows the estimated relative current and R&D energy savings opportunities for individual processes based on the sector-wide extrapolated data.

The U.S. iron and steel industry operated at relatively low capacity utilization and lower-thantypical efficiencies in 2010, due in large part to the economic downturn. While the specific impacts of the economic factors in 2010 are not directly identified in this report, it is reasonable to assume that the current opportunity is likely somewhat exaggerated, as a portion of the current savings could be achieved by simply optimizing production rates. For this reason the border between current opportunity and R&D opportunity is not explicitly defined, and a dashed line and color fading is used in Figure ES-1.

Table ES-1. Potential Energy Savings Opportunities in the U.S. Iron and Steel Manufacturing Sector <sup>[1]</sup>		
Opportunity Bandwidths	Estimated Energy Savings Opportunity for Six Select Iron and Steel Manufacturing Processes (per year)	Estimated Energy Savings Opportunity for All of the U.S. Iron and Steel Sector Based on Extrapolated Data (per year)
<i>Current Opportunity</i> – energy savings if the best technologies and practices available are used to upgrade production	197 TBtu <sup>2</sup> (39% energy savings, where TM is the baseline)	240 TBtu <sup>3</sup> (39% energy savings, where TM is the baseline)
<i>R&amp;D Opportunity</i> – additional energy savings if the applied R&D technologies under development worldwide are deployed	124 TBtu <sup>4</sup> (24% energy savings, where TM is the baseline)	150 TBtu <sup>5</sup> (24% energy savings, where TM is the baseline)

 $^{2}$  197 TBtu = 822 - 625

<sup>&</sup>lt;sup>1</sup> The energy estimates presented in this study are for macro-scale consideration; energy intensities and energy consumption values do not represent energy use in any specific facility or any particular region in the United States. The costs associated with achieving energy savings are not considered in this study. All estimates are for onsite energy use (i.e., energy consumed within the refinery boundary). Energy used as feedstocks (non-fuel inputs) to production is excluded.

 $<sup>^{3}</sup>$  240 TBtu = 999 - 759

 $<sup>^{4}</sup>$  124 TBtu = 625 - 501

 $<sup>^{5}</sup>$  150 TBtu = 759 - 609

vi Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing





The PM energy consumption estimates are speculative because they are based on unproven technologies. The estimates assume deployment of R&D technologies that are under development; where multiple technologies were considered for a similar application, only the most energy efficient technology was considered in the energy savings estimate. The difference between PM and TM is labeled "impractical" because with today's knowledge of technologies in R&D, further investment may no longer lead to incremental energy savings and thermodynamic limitations impede technology opportunities.

The results presented show that 197 TBtu of energy could be saved each year if capital investments in the best technologies and practices available worldwide are used to upgrade six iron and steel manufacturing processes; an additional 124 TBtu could be saved through the adoption of applied R&D technologies under development worldwide.

However, if the energy savings potential is estimated for the U.S. iron and steel industry as a whole, the current energy savings opportunity is 240 TBtu per year and the R&D opportunity increases to 150 TBtu per year.

The top four Current Energy Savings Opportunities for the processes are as follows:

- Hot rolling 83 TBtu (or 35% of current opportunity)
- Cold rolling 47 TBtu (or 20% of current opportunity)
- All other NAICS 331111<sup>6</sup> processes 42 TBtu (or 18% of the current opportunity)
- Basic oxygen furnace (BOF) steelmaking 30 TBtu (or 13% of the current opportunity).

The top four R&D Energy Saving Opportunities for the processes are as follows:

- Ironmaking 49 TBtu (or 32% of the R&D opportunity)
- Hot rolling 30 TBtu (or 20% of the R&D opportunity)
- All other NAICS 331111<sup>7</sup> processes 27 TBtu (or 18% of the R&D opportunity)
- Electric arc furnace (EAF) steelmaking- 21 TBtu (or 14% of the R&D opportunity),

<sup>&</sup>lt;sup>6</sup> All other NAICS 331111 includes all other processes in the iron and steel sector other than the six processes studied.

<sup>&</sup>lt;sup>7</sup> All other NAICS 331111 includes all other processes in the iron and steel sector other than the six processes studied.

viii Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

## **List of Acronyms and Abbreviations**

AISI	American Iron and Steel Institute
AMO	Advanced Manufacturing Office
AOD	Argon oxygen decarburization
BF	Blast furnace
BFG	Blast furnace gas
BOF	Basic oxygen furnace
Btu	British thermal unit
COE	Cost of energy
COG	Coke oven gas
CT	Current typical energy consumption or energy intensity
DOE	U.S. Department of Energy
DRI	Direct-reduced iron
EAF	Electric arc furnace
EERE	DOE Office of Energy Efficiency and Renewable Energy
EIA	U.S. Energy Information Administration
EPA	U.S. Environmental Protection Agency
GJ	Gigajoules
HHV	Higher heating value
IEA	International Energy Agency
Κ	Kelvin
kWh	Kilowatt hours
LHV	Lower heating value
MECS	Manufacturing Energy Consumption Survey
mm	Millimeter
MMBtu	Million British thermal units
MMBtu/ton	Million British thermal units per short ton
MT	Metric ton (tonne)
NAICS	North American Industry Classification System
PJ	Petajoules
PM	Practical minimum energy consumption or energy intensity
SOA	State of the art energy consumption or energy intensity
TBtu	Trillion British thermal units
ТМ	Thermodynamic minimum energy consumption or energy intensity

## **Table of Contents**

Pre	eface		iii
Ex	ecutiv	ve Summary	v
Lis	st of A	Acronyms and Abbreviations	ix
1.	1.1. 1.2. 1.3.	duction Overview Comparison to Other Bandwidth Studies Definitions of Energy Consumption Bands and Opportunity Bandwidths Bandwidth Analysis Method	1 1 2
2.	U.S.	Iron and Steel Manufacturing Sector Overview	7
		U.S. Iron and Steel Manufacturing Economic Overview	
		U.S. Iron and Steel Manufacturing Products, Establishments, and Processes	
	2.2.	2.2.1. Ore Agglomeration	
		2.2.2. Cokemaking	
		2.2.3. Ironmaking	
		2.2.3.1. Blast Furnace Ironmaking	
		2.2.3.2. Direct Reduction Ironmaking	
		2.2.4. Steelmaking	
		2.2.4.1. Basic Oxygen Furnace Steelmaking	
		2.2.4.2. Electric Arc Furnace Steelmaking	
		2.2.5. Casting	
		2.2.6. Rolling	
	2.3	U.S. Iron and Steel Manufacturing Energy Consumption	
	2.3.	2.3.1. Fuel and Feedstocks	
		2.3.2. Electricity	
		2.3.3. Steam	
3.	Prod	uction Volumes in U.S. Iron and Steel Manufacturing	
4.	Curr	ent Typical Energy Consumption for U.S. Iron and Steel	
		ufacturing	24
		Boundaries of the Iron and Steel Bandwidth Study	
		Estimated Energy Intensity for Individual Processes	
		4.2.1. Ore Agglomeration	
		4.2.2. Cokemaking	
		4.2.3. Ironmaking	
		4.2.4. Steelmaking	
		4.2.5. Casting and Rolling	

	4.3.	Calculated Current Typical Energy Consumption for Individual Processes	
	4.4.	Current Typical Energy Consumption By Process and Sector-wide	30
5.	State	e of the Art Energy Consumption for U.S. Iron and Steel	
		ufacturing	32
		Calculated State of the Art Energy Consumption for Individual Processes	
		5.1.1. Ore Agglomeration	
		5.1.2. Cokemaking	
		5.1.3. Ironmaking	
		5.1.4. Steelmaking	
		5.1.5. Casting and Rolling	34
		5.1.6. Other Estimates of SOA Savings	35
	5.2.	State of the Art Energy Consumption By Process and Sector-wide	35
		5.2.1. Comparing State of the Art and Current Typical Energy Data	
6.	Prac	tical Minimum Energy Consumption for U.S. Iron and Steel	
	Man	ufacturing	42
	6.1.	R&D in the Iron and Steel Industry	42
	6.2.	Calculated Practical Minimum Energy Consumption for Individual Processes	43
		6.2.1. Weighting of Technologies	45
	6.3.	Practical Minimum Energy Consumption By Process and Sector-wide	46
7.	Ther	modynamic Minimum Energy Consumption for U.S. Iron and Steel	
7.		modynamic Minimum Energy Consumption for U.S. Iron and Steel	52
7.	Man		
7.	<b>Man</b> 7.1.	ufacturing	
7.	<b>Man</b> 7.1.	ufacturing	52
7.	<b>Man</b> 7.1.	Ufacturing Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual	52
7.	<b>Man</b> 7.1.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes 7.2.1. Ore Agglomeration and Cokemaking 7.2.2. Ironmaking	52 52 53 53
7.	<b>Man</b> 7.1.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes 7.2.1. Ore Agglomeration and Cokemaking 7.2.2. Ironmaking 7.2.3. Steelmaking	52 52 53 53 54
7.	<b>Man</b> 7.1.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes 7.2.1. Ore Agglomeration and Cokemaking 7.2.2. Ironmaking 7.2.3. Steelmaking 7.2.4. Casting and Rolling	52 52 53 53 54 54
7.	<b>Man</b> 7.1.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes	52 52 53 53 54 54
	Man 7.1. 7.2. 7.3.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes 7.2.1. Ore Agglomeration and Cokemaking 7.2.2. Ironmaking 7.2.3. Steelmaking 7.2.4. Casting and Rolling	52 52 53 53 54 54 54 54
	Man 7.1. 7.2. 7.3. <b>U.S.</b>	<b>ufacturing</b> Thermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide	
8.	Man 7.1. 7.2. 7.3. <b>U.S.</b> 8.1.	Thermodynamic Minimum Energy Calculated Thermodynamic Minimum Energy Consumption for Individual Processes 7.2.1. Ore Agglomeration and Cokemaking 7.2.2. Ironmaking 7.2.3. Steelmaking 7.2.4. Casting and Rolling Thermodynamic Minimum Energy Consumption By Process and Sector-wide Iron and Steel Manufacturing Energy Bandwidth Summary	52 52 53 54 54 54 54 54 54
8.	Man 7.1. 7.2. 7.3. U.S. 8.1. Refe	ufacturing   Thermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide   Iron and Steel Manufacturing Energy Bandwidth Summary   Iron and Steel Bandwidth Profile	52 52 53 53 54 54 54 56 56
8. 9. Ap	Man 7.1. 7.2. 7.3. U.S. 8.1. Refe	<b>Ufacturing</b> Thermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide   Iron and Steel Manufacturing Energy Bandwidth Summary   Iron and Steel Bandwidth Profile   rences   ix A1: Master Iron and Steel Table	52 52 53 53 54 54 54 56 56
8. 9. Ap	Man 7.1. 7.2. 7.3. U.S. 8.1. Refe	ufacturing   Thermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide   Iron and Steel Manufacturing Energy Bandwidth Summary   Iron and Steel Bandwidth Profile   rences   ix A1: Master Iron and Steel Table   ix A2: References for U.S. Production Data of the 6 Processes	52 52 53 53 54 54 54 56 56
8. 9. Ap	Man 7.1. 7.2. 7.3. U.S. 8.1. Refe opend Stud	Intermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide   Iron and Steel Manufacturing Energy Bandwidth Summary   Iron and Steel Bandwidth Profile   rences   ix A1: Master Iron and Steel Table   ix A2: References for U.S. Production Data of the 6 Processes   ied and Energy Intensity Data Used to Calculate the Current	52 52 53 53 54 54 54 56 56
8. 9. Ap	Man 7.1. 7.2. 7.3. U.S. 8.1. Refe opend Stud Typi	ufacturing   Thermodynamic Minimum Energy   Calculated Thermodynamic Minimum Energy Consumption for Individual   Processes   7.2.1. Ore Agglomeration and Cokemaking   7.2.2. Ironmaking   7.2.3. Steelmaking   7.2.4. Casting and Rolling   Thermodynamic Minimum Energy Consumption By Process and Sector-wide   Iron and Steel Manufacturing Energy Bandwidth Summary   Iron and Steel Bandwidth Profile   rences   ix A1: Master Iron and Steel Table   ix A2: References for U.S. Production Data of the 6 Processes	52 52 53 53 54 54 56 60 65

Appendix A3: CT, SOA, and TM Energy Intensities by Process and Energy	
Туре	67
Current Typical Energy Intensities	67
State of The Art Energy Intensities	71
Theoretical Minimum Energy Intensities	75
Appendix A4: Technologies Analyzed to Estimate Practical Minimum	
Energy Intensities with References	80
References for Table A4	85
Appendix A5: Practical Minimum Technology Weighting Factors	92
Methodology to Determine Weighting Factors	92

## **List of Figures**

Figure ES-1. Current and R&D Energy Savings Opportunities for the Six Processes
Studied and for Iron and Steel Sector-wide Based on Extrapolated Data vii
Figure 1-1. Energy Consumption Bands and Opportunity Bandwidths Estimated in this
Study
Figure 2-1. Geographic Distribution of Integrated and Mini Steel Mills in the United
States (EPA 2011)
Figure 2-2. Steelmaking Flowlines for Integrated and Mini Mills (AISI 2013b) 10
Figure 2-3. Onsite Energy Entering U.S. Iron and Steel Mills, 2010 (DOE 2014) 15
Figure 2-4. Onsite Energy Consumption at Point of End Use in U.S. Iron and Steel Mills,
2010 ( <i>DOE 2014</i> )
Figure 2-5. Fuel Consumption in the Iron and Steel Sector by End Use, 2010 (DOE 2014) 17
Figure 2-6. Feedstock Energy Consumption in the Iron and Steel Sector, 2010 (EIA
2013)
Figure 2-7. Electricity Consumption in the Iron and Steel Sector by End Use, 2010 (DOE
2014)
Figure 2-8. Steam Consumption in the Iron and Steel Sector by End Use, 2010 (DOE
2014)
Figure 5-1. Current Opportunity Energy Savings Bandwidths for Processes Studied (with
Percent of Overall Energy Consumption Bandwidth)
Figure 5-2. Current Energy Savings Opportunity by Process
Figure 5-3. Steel energy intensity between 1990 and 2010 (AISI 2012)
Figure 6-1. Steel Triangle Processing Route Framework (Sadoway 2008)
Figure 6-1. Current and R&D Opportunity Energy Savings Bandwidths for the Iron and
Steel Processes Studied (with Percent of Overall Energy Consumption
Bandwidth)
Figure 6-2. Current and R&D Energy Savings Opportunities by Iron and Steel Process
Studied (Energy Savings per Year in TBtu)
Figure 8-1. Current and R&D Energy Savings Opportunities in U.S. Iron and Steel for the
Processes Studied and for Sector-Wide based on Extrapolated Data
Figure 8-2. Current and R&D Opportunity Energy Savings Bandwidths for the Processes
Studied (with Percent of Overall Energy Consumption Bandwidth)

#### **List of Tables**

Table ES-1. Potential Energy Savings Opportunities in the U.S. Iron and Steel	
Manufacturing Sector <sup>[1]</sup>	vi
Table 2-1. Iron and Steel Process Areas Considered in the Bandwidth Analysis	10
Table 2-2. U.S. Iron and Steel Manufacturing Energy Consumption Sector-Wide, 2010	14
Table 3-1. U.S. Iron and Steel Production Volumes for Processes Studied in 2010	22
Table 4-1. Published Sources Reviewed to Identify Current Typical Energy Intensities for	
Processes Studied	25
Table 4-2. Calculated U.S. Onsite Current Typical Energy Consumption for Processes	
Studied in 2010 with Calculated Primary Energy Consumption and Offsite	
Losses	29
Table 4-3. Onsite and Primary Current Typical Energy Consumption for the Nine	
Processes Studied and Sector-Wide in 2010, with Percent of Sector	
Coverage	31
Table 5-1. Published Sources Referenced to Identify State of the Art Energy Intensities	
for Nine Select Processes	33
Table 5-2. Onsite State of the Art Energy Consumption, Energy Savings, and Energy	
Savings Percent for the Processes Studied and Sector-Wide	37
Table 6-2. Onsite Practical Minimum Energy Consumption, Energy Savings, and Energy	
Savings Percent for the Processes Studied and Sector-Wide	47
Table 7-1. Published Sources Reviewed to Identify Thermodynamic Minimum Energy	
Intensities for the Processes Studied	52
Table 7-2. Thermodynamic Minimum Energy Consumption by Process and Sector-Wide	
for the Six Processes Studied and Extrapolated to Sector Total	55
Table 8-1. Current Opportunity and R&D Opportunity Energy Savings for the Nine	
Processes Studied and Extrapolated to Sector-Wide Total	57
Table A1. U.S. Production Volume of Six Iron and Steel Processes in 2010 with Energy	
Intensity Estimates and Calculated Onsite Energy Consumption for the	
Four Bandwidth Measures (Excludes Feedstock Energy)	65
Table A2. References for U.S. Throughput Data of the Six Iron and Steel Processes	
Studied and Energy Intensity Data Used to Calculate the Current Typical,	
State of the Art, and Thermodynamic Minimum Energy Consumption	
Bands	
Table A3-1. Current Typical Energy Intensity Estimate for Pelletizing	
Table A3-2. Current Typical Energy Intensity Estimate for Sintering	
Table A3-3. Current Typical Energy Intensity Estimate for Cokemaking	
Table A3-4. Current Typical Energy Intensity Estimate for Blast Furnace Ironmaking	68
Table A3-5. Current Typical Energy Intensity Estimate for Basic Oxygen Furnace	_
Steelmaking	69
Table A3-6. Current Typical Energy Intensity Estimate for Electric Arc Furnace	
Steelmaking	69

Table A3-7. Current Typical Energy Intensity Estimate for Direct Reduced Ironmaking
Table A3-8. Current Typical Energy Intensity Estimate for Continuous Casting
Table A3-9. Current Typical Energy Intensity Estimate for Hot Rolling
Table A3-10. Current Typical Energy Intensity Estimate for Cold Rolling
Table A3-11. State of the Art Energy Intensity Estimate for Sintering
Table A3-12. State of the Art Energy Intensity Estimate for Cokemaking
Table A3-13. State of the Art Energy Intensity Estimate for Blast Furnace Ironmaking
Table A3-14. State of the Art Energy Intensity Estimate for Basic Oxygen Furnace
Steelmaking72
Table A3-15. State of the Art Energy Intensity Estimates for Electric Arc Furnace
Steelmaking72
Table A3-16. State of the Art Energy Intensity Estimate for Direct Reduced Ironmaking72
Table A3-17. State of the Art Energy Intensity Estimate for Continuous Casting
Table A3-18. State of the Art Energy Intensity Estimate for Cold Rolling Mill73
Table A3-19. State of the Art Energy Intensity Estimate for Hot Rolling74
Table A3-20. Theoretical Minimum Energy Intensity Estimate for Sintering
Table A3-21. Theoretical Minimum Energy Intensity Estimate for Cokemaking75
Table A3-22. Theoretical Minimum Energy Intensity Estimate for Blast Furnace
Ironmaking75
Table A3-23. Theoretical Minimum Energy Intensity Estimate for Basic Oxygen Furnace
Steelmaking75
Table A3-24. Theoretical Minimum Energy Intensity Estimate for Electric Arc Furnace
Steelmaking76
Table A3-25. Theoretical Minimum Energy Intensity to Produce Direct Reduced Iron at
1173K (900°C or 1620°F) Reduction Temperature for Selected Conditions 76
Table A3-26. Theoretical Minimum Energy Intensity to Roll Steel for Selected Products
and Conditions77
Table A3-27. Conversion from Lower Heating Value to Higher Heating Value for
Table A3-27. Conversion from Lower Heating Value to Higher Heating Value for   Current Typical Energy Intensity
Current Typical Energy Intensity
Current Typical Energy Intensity

xvi Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

# **1.** Introduction

## 1.1. OVERVIEW

This bandwidth study examines energy consumption and potential energy savings opportunities in the U.S. iron and steel manufacturing sector, as defined by classification 331111 of the North American Industry Classification System (NAICS). The purpose of this data analysis is to provide macro-scale estimates of energy savings opportunities for iron and steel manufacturing processes and iron and steel sector-wide. In this study, four different energy consumption *bands* (or measures) are estimated. The *bandwidth*—the difference between bands of energy consumption—is the estimated potential energy savings opportunity.

The United States produces a wide range of steel products using various processes; six of the most energy-intensive iron and steel manufacturing processes were studied. Together, these processes accounted for 82% of the onsite energy consumption of the U.S. iron and steel sector in 2010.

The four bands of energy consumption estimated in this report include: the onsite energy consumption associated with six iron and steel manufacturing processes in 2010 (current typical); two hypothetical energy consumption levels with progressively more advanced technologies and practices (state of the art and practical minimum); and one energy consumption level based on the minimum amount of energy needed to theoretically complete an iron and steel manufacturing process (thermodynamic minimum). The bands of energy consumption are used to calculate *current* and *R&D opportunity* bandwidths for energy savings.

## **1.2. COMPARISON TO OTHER BANDWIDTH STUDIES**

This study builds upon the 2004 DOE bandwidth report *Steel Industry Energy Bandwidth Study*. This study compares diverse industrial, academic and governmental consumption data to analyses<sup>1</sup> of reported plant energy consumption data in the Manufacturing Energy Consumption Survey (MECS) conducted by the U.S. Energy Information Administration (EIA) for data year 2010. This study also updates energy consumption and production values to the year 2010.

This report is one in a series of bandwidth studies commissioned by DOE's Advanced Manufacturing Office characterizing energy consumption in U.S. manufacturing using a uniform methodology and definitions of energy bands. Other manufacturing sector bandwidth studies include chemicals, petroleum refining, and pulp and paper; additional sector studies are under consideration. Collectively, these studies explore the potential energy savings opportunities in

<sup>&</sup>lt;sup>1</sup> The relevant analysis was published as the *Manufacturing Energy and Carbon Footprint for the Iron and Steel Sector* (NAICS 3311, 3312), based on energy use data from 2010 EIA MECS (with adjustments) in February 2014. Hereafter, this document will be referred to as the "Energy Footprint" and listed in the References section as DOE 2014.

manufacturing that are available through existing technology and with investment in research and development (R&D) technologies.

## 1.3. DEFINITIONS OF ENERGY CONSUMPTION BANDS AND OPPORTUNITY BANDWIDTHS

There are four energy consumption bands referenced throughout this report: current typical (CT), state of the art (SOA), practical minimum (PM), and thermodynamic minimum (TM) energy consumption. These bands describe different levels of energy consumption for iron and steel manufacturing processes.

As shown in Figure 1-1, the bands progress from higher to lower levels of energy consumption, reflecting the use of increasingly more efficient manufacturing technologies and practices. The upper bound is set by a mix of new and older technologies and practices in current use (the current typical level of energy consumption). The lower bound is defined by the theoretical minimum energy requirement assuming ideal conditions and zero energy losses (the thermodynamic minimum level of energy consumption).

Each of these two bounds defining the extremes of energy consumption can be compared to hypothetical measures in the middle of this range. If manufacturers use the most



efficient technologies and practices available in the world, energy consumption could decrease from the current typical to the level defined by the state of the art. Since these state of the art technologies already exist, the difference between the current typical and the state of the art energy consumption levels defines the *current opportunity* to decrease energy consumption. Given that this is an evaluation of technical potential, fully realizing the current opportunity would require investments in capital that may or not be economically viable for any given facility.

Widespread deployment of future R&D advanced technologies and practices under investigation by researchers around the globe could help manufacturers attain the practical minimum level of

energy consumption. The difference between state of the art and practical minimum levels of energy consumption defines the *R&D opportunity* for energy savings.

Definitions of the four energy bands are provided in the inset (box at right). Definitions of the two opportunity bandwidths are provided below:

The *current opportunity* is the energy savings that is potentially attainable through capital investments in the best technologies and practices available worldwide. It is the difference between CT and SOA energy consumption.

The *R&D opportunity* is the energy savings that is potentially attainable through the applied R&D technologies under development. It is the difference between SOA and PM energy consumption. To attain this energy savings, iron and steel mills would need to manufacture products in new ways with technologies that are not commercially available.

#### Definitions of Energy Bands Used in the Bandwidth Studies

The following definitions are used to describe different levels of U.S. energy consumption for a *specific manufacturing process industry-wide*:

**Current Typical (CT) energy consumption:** U.S. energy consumption in 2010.

**State of the Art (SOA) energy consumption:** The minimum amount of energy required assuming the adoption of the best technologies and practices available worldwide.

**Practical Minimum (PM) energy consumption:** The minimum amount of energy required assuming the deployment of the best applied R&D technologies under development worldwide.

This measure is expressed as a range to reflect the speculative nature of the energy impacts of the unproven technologies considered.

**Thermodynamic Minimum (TM) energy consumption:** The minimum amount of energy theoretically required assuming ideal conditions typically unachievable in realworld applications.

The difference between PM and TM energy consumption is labeled as *impractical*. The term *impractical* is used because with today's knowledge of technologies in R&D, further investment may no longer lead to incremental energy savings and thermodynamic limitations impede technology opportunities.

### **1.4. BANDWIDTH ANALYSIS METHOD**

This Section describes the method used in this bandwidth study to estimate the four bands of energy consumption and the two corresponding energy savings opportunity bandwidths. This section can also be used as a guide to understanding the structure and content of this report.

In this study, U.S. energy consumption is labeled as either "onsite energy" or "primary energy" and defined as follows:

• **Onsite energy** (sometimes referred to as site or end use energy) is the energy consumed within the manufacturing plant boundary (i.e., within the plant gates). Non-fuel feedstock energy is *not* included in the onsite energy consumption values presented in this study.

• **Primary energy** (sometimes referred to as source energy) includes energy that is consumed both offsite and onsite during the manufacturing process. Offsite energy consumption includes generation and transmission losses associated with bringing electricity and steam to the plant boundary. Non-fuel feedstock energy is not included in the primary energy values. Primary energy is frequently referenced by governmental organizations when comparing energy consumption across sectors.

Four bands of energy consumption are quantified for select individual processes and iron and steel manufacturing sector-wide. **The bands of energy consumption and the opportunity bandwidths presented herein consider onsite energy consumption; feedstocks<sup>2</sup> are excluded.** To determine the total annual onsite CT, SOA, PM, and TM energy consumption values of the processes studied (TBtu per year), energy intensity values per unit weight (Btu per pound of product) are estimated and multiplied by the production volumes (pounds per year of product). The year 2010 is used as a base year since it is the most recent year for which consistent sector-wide energy consumption data are available. Unless otherwise noted, 2010 production data is used. Some iron and steel manufacturing processes are exothermic and are net producers of energy; the net energy was considered in the analysis.

The estimates presented are for macro-scale consideration of energy use in iron and steel manufacturing. The estimates reported herein are representative of average U.S. iron and steel manufacturing; they do not represent energy use in any specific facility or any particular region in the United States or the world.

Significant investment in technology development and implementation would be needed to fully realize the potential energy savings opportunities estimated. The costs associated with achieving SOA and PM energy consumption are not considered in this report; a techno-economic analysis of the costs and benefits of future R&D technologies was not in the scope of this study.

The calculated energy consumption values in this report are based on an examination of referenced data and extrapolation to sector-wide energy savings opportunities. The references, methodology, and assumptions employed are presented with the data in each chapter and were peer reviewed.

**Overview of energy use in iron and steel manufacturing:** Chapter 2 provides an **overview** of the U.S. iron and steel sector and how energy is used in iron and steel manufacturing (how much, what type, and for what end uses).

*Estimating production volumes for select processes:* Chapter 3 presents the relevant **production volumes** for the six processes (tons per year) in 2010 and the rationale for how the six processes were selected.

<sup>&</sup>lt;sup>2</sup> Feedstock energy is the nonfuel use of combustible energy. Feedstocks are converted to iron and steel products (not used as a fuel); MECS values reported as "feedstocks" exclude feedstocks converted to other energy products.

<sup>4</sup> Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

*Estimating CT energy consumption:* Chapter 4 presents the calculated onsite **CT energy consumption** (TBtu per year) for the six processes individually and sector-wide (along with references for the CT energy intensity data and assumptions). The CT energy consumption data is calculated based on this energy intensity data and the production volumes (identified in Chapter 3). The boundary assumptions for the industrial processes considered in this bandwidth study are presented.

MECS provides onsite CT energy consumption data sector-wide for 2010 (See Table 4-1). However, MECS does not provide CT energy consumption data for individual processes. The percent coverage of the processes studied (compared to MECS sector-wide data) is presented and used in calculations discussed later in this report.

Primary CT energy consumption (TBtu per year) estimates are calculated, which include offsite generation and transmission losses associated with bringing electricity and steam to manufacturing facilities. Primary energy consumption estimates are not provided for SOA, PM, or TM because they were outside the scope of this study.

*Estimating SOA energy consumption*: Chapter 5 presents the estimated onsite **SOA energy consumption** for the six processes (along with the references for the SOA energy intensity data and assumptions). The sector-wide SOA energy consumption is estimated based on an extrapolation of the SOA energy consumption for the six processes studied. The *current opportunity* bandwidth, the difference between CT energy consumption and SOA energy consumption (also called the SOA energy savings), is presented along with the SOA energy savings percent.

**Estimating PM energy consumption:** Chapter 6 presents the estimated onsite **PM energy consumption** for the six processes (along with the references for PM energy intensity data and assumptions). The range of potentially applicable applied R&D technologies to consider in the PM analysis worldwide is vast. The technologies that were considered are sorted by process and described in Appendix A4. The technologies that are considered crosscutting throughout all of iron and steel manufacturing along with the most energy-saving, process-specific R&D technologies were used to determine PM energy consumption for each process. A weighting method that includes factors such as technology readiness, cost, and environmental impact was developed for all technologies considered; the weighting analysis methodology and summary table provided in Appendix A5 is intended to serve as a resource for continued consideration of all identified R&D opportunities.

The sector-wide PM energy consumption is estimated based on an extrapolation of the PM energy consumption for the six processes studied. The *R&D opportunity* bandwidth, the difference between SOA energy consumption and PM energy consumption, is presented along with the PM energy savings percent. PM energy savings is the sum of *current* and *R&D opportunity*.

The technologies considered in the PM analysis are unproven on a commercial scale. As a result, the PM energy consumption is expressed as a range. The upper limit is assumed to be the SOA energy consumption; the lower limit is estimated and shown as a dashed line with color fading in the summary figures because the PM is speculative and depends on unproven R&D technologies. Furthermore, the potential energy savings opportunity could be greater if additional unproven technologies were considered.

*Estimating TM energy consumption:* Chapter 7 presents the estimated onsite **TM energy consumption** for the six processes (along with the references for the TM energy intensity data and assumptions). The TM energy intensities are based on the commercial process pathways. TM energy consumption assumes all of the energy is used productively and there are no energy losses. TM is the minimum amount of energy required; in some cases it is less than zero.

To determine the potential available energy savings opportunities in this bandwidth study, TM energy consumption was used as the baseline for calculating the energy savings potentials for each process studied (not zero, as is typically the case in considering energy savings opportunities). The rationale for using TM as the baseline is explained in Chapter 7.

*Estimating the energy savings opportunities:* Chapter 8 presents the energy savings **opportunity bandwidths** for the processes and sector-wide. The analyses used to derive these values are explained in Chapters 3 to 7.

## 2. U.S. Iron and Steel Manufacturing Sector Overview

This Chapter presents an overview of the U.S. iron and steel manufacturing sector, including its impact on the economy and jobs, number of establishments, types of energy consumed, and the end uses of the energy. The convention for reporting energy consumption as either onsite versus primary energy is explained. The data and information in this Chapter provide the basis for understanding the energy consumption estimates.

## 2.1. U.S. IRON AND STEEL MANUFACTURING ECONOMIC OVERVIEW

Steel is a vital domestic manufacturing product and is important for many applications including construction (residential, commercial, transportation), automotive, machinery and equipment, containers, and national security, among others.

In 2010, the U.S. produced 88.7 million short tons (referred to as tons in the document unless otherwise noted) of raw steel while operating at about 70.4% of estimated capacity (AISI 2011a). The U.S. was the third largest steel manufacturing country in 2010 behind China and Japan, accounting for 5.7% of world steel production (AISI 2011a). Also in 2010, the U.S. steel industry as a whole contributed \$17.5 billion in GDP (Considine 2012). The value of steel shipped in the U.S. in 2010 totaled \$70 billion and weighed 84 million tons (AISI 2013a).

Domestic steelmakers have improved the energy intensity per ton of steel produced by 27% between 1990 and 2010 (AISI 2012). This reduction in energy consumption and corresponding increase in production is largely due to the shift from ore-based to scrap-based steel production, although deployment of new energy efficient steelmaking technologies and innovations on the plant floor are also contributors to this improvement. Steel from a variety of products such as containers, automobiles, and appliances can be recycled once they reach the end of their useful life. In fact, steel is the most recycled material in North America, with more steel recycled annually than paper, aluminum, glass, and plastic combined (SRI 2013a). In 2012, the steel recycling rate was 88%, up from a rate of 64% in 2000 (SRI 2013b). There continues to be further savings opportunity in this energy-intensive sector through both energy consumption and production efficiency improvements.

## 2.2. U.S. IRON AND STEEL MANUFACTURING PRODUCTS, ESTABLISHMENTS, AND PROCESSES

Iron and steel operations are complex and large facilities that produce significant quantities of steel each year. Steel mills in the U.S. are generally either integrated mills, or mini mills; the primary difference being the proportion of recycled steel that is used (up to 99% in mini mills and was 91% in 2010 (USGS 2012c)). The distribution of many of the integrated and mini mills in the United States is shown in Figure 2-1. Integrated steel mills produce steel from iron ore via

the blast furnace (BF) and basic oxygen furnace (BOF) steelmaking technology while mini steels mills produce steel mostly from recycled scrap steel via the electric arc furnace (EAF) steelmaking technology. Figure 2-2 shows the steel industry process flowlines for integrated mills and for mini mills.

In 2010, there were about 15 BF/BOF steelmaking facilities operated by five companies and 112 EAF steelmaking facilities operated by over 50 companies in the U.S. (USGS 2012a). Most of these steelmaking facilities in the U.S. are concentrated in Indiana, Ohio, Pennsylvania, Michigan, and Illinois due to the close proximity to coal and iron ore suppliers among other factors. In 2010, iron and steel manufacturing directly employed 135,000 workers and total employment (including both direct and indirect employees in other industries) was estimated at 1,080,000 (AISI 2013a).



Figure 2-1. Geographic Distribution of Integrated and Mini Steel Mills in the United States (EPA 2011)





Figure 2-2. Steelmaking Flowlines for Integrated and Mini Mills (AISI 2013b)

This study focuses on energy consumption in six energy intensive process areas in steel manufacturing. These process areas are identified in Table 2-1, along with some of the major sub processes. Energy intensity and consumption is evaluated by process area and sub process for CT, SOA, PM, and TM in Sections 4 through 7 of this report. CT energy intensity for pelletizing iron ore is shown for reference purpose only because this sub process is outside the boundary of bandwidth analysis. Although direct reduction ironmaking was not used in the United States in 2010, and therefore not included in the bandwidth savings summary, it is likely to appear in future steel industry energy analyses and so energy intensity estimates have been included. CT, SOA and TM energy intensity for direct reduction is included for reference only in this study.

Table 2-1. Iron and Steel Process Areas Considered in the Bandwidth Analysis		
Processes	Sub Processes	
Agglomeration	Sintering, Pelletizing	
Cokemaking		
Ironmaking	Blast Furnace, Direct Reduction	
Steelmaking	Basic Oxygen Furnace, Electric Arc Furnace	
Casting		
Rolling	Hot Rolling, Cold Rolling	

There are two main processes for producing steel: integrated steelmaking, which combines a blast furnace with a BOF, and EAF steelmaking. These two processes are distinctly different as the integrated BOF process consumes mostly agglomerated iron ore along with some scrap steel (up to 30%; was 24% in 2010 (USGS 2012c)) while the EAF process consumes mostly scrap steel as well as reduced iron, cast iron, and other iron containing materials to produce raw steel (WCA 2013). It requires about seven times the amount of energy to produce a ton of steel from ore in a blast furnace and BOF (including the energy for cokemaking, pelletizing, and sintering), compared to remelting scrap in an electric arc furnace (not including losses for generating and transmitting electricity) (IPPC 2013; LBNL 2008). However, many other factors come in to play in the economics of ore-based versus scrap-based steelmaking. For this reason, it is useful to consider the two production pathways separately in this bandwidth study.

#### 2.2.1. Ore Agglomeration

Ore agglomeration is used to enhance the iron content and physical properties of low-grade iron ore that will be used for ironmaking. The two main processes associated with ore agglomeration are sintering and pelletizing. Sintering involves mixing ore pellets and fines, coke breeze or coal fines, and lime (which acts as both a flux and a binder), and then heating to form hardened and porous lumps of sinter. This mixture improves the efficiency of the reduction reaction in the blast furnace and the physical shape and strength of the sinter provides support and open channels for gas flow in the vessel. Sintering is an energy intensive step in the steelmaking due to the process heating.

Pelletizing involves crushing and grinding the iron ore to remove impurities and then the forming of round, uniform size iron ore pellets and heating to harden them. Pelletizing is usually conducted near the ore mining site, and not at the steel mill. Briquetting is another type of agglomeration process, which involves the formation of iron ore briquettes from iron ore and fines and hardening at elevated temperatures. A total of 45.3 million tons of iron ore and agglomerated products were consumed in blast furnaces in 2010, while only 0.6 million tons were consumed in steelmaking furnaces. The total 45.9 million tons was composed of 86% pellets, 13% sinter, briquettes, nodules, and other, and 1% natural ore (AISI 2011a).

### 2.2.2. Cokemaking

Coke is an important and necessary raw material for ironmaking in a blast furnace. The process of cokemaking is an energy intensive process that involves charging coal into a coke oven, where it is heated to high temperatures in an airless environment. This drives off volatile chemicals, thus increasing the purity of the carbon. The heating process is self-sustaining and is fueled by oxidation of some of the coal. Coke oven off-gas contains high amounts of carbon monoxide which is usually recovered for its calorific value. Electricity can be produced through the combustion of coke oven off-gases to provide power for other plant operations. In 2010, integrated steel plants in the U.S. produced 9.3 million tons of coke and consumed 12 million tons (AISI 2011a). Coke is also produced at merchant plants.

#### 2.2.3. Ironmaking

Ironmaking is the process where iron ore is converted to iron.

#### 2.2.3.1. Blast Furnace Ironmaking

Traditional Ironmaking takes place in a blast furnace. Iron ore agglomerate, a limestone flux, and coke are fed into the furnace where the coke is reacted with preheated air in order to heat and reduce (remove oxygen from) the iron ore resulting in a high carbon molten iron product known as pig iron. The quality of the coke that is used affects energy use in the blast furnace; the lower the quality of coke, the greater the volume of coke required. Pig iron produced in the blast furnace typically contains up to 5% by weight carbon, up to 2.5% by weight manganese, up to 4% by weight silicon, and phosphorus and sulfur (Fruehan et al. 2000; McHannon 1971).

Before being tapped from the furnace, pig iron collects in a molten pool with a protective layer of slag (principally calcium oxide from the lime and silicon oxide native to the ore). After tapping, which occurs approximately every eight hours, the pig iron is typically routed to a basic oxygen furnace where oxygen is blown through the molten metal to oxidize impurities and produce steel. About 29.6 million tons of pig iron was produced in the U.S. in 2010, all of which was consumed for steelmaking (AISI 2011a). Globally, the U.S. ranked 8<sup>th</sup> in pig iron production and accounted for 3% of world production (USGS 2012b).

#### 2.2.3.2. Direct Reduction Ironmaking

An alternative ore-based ironmaking process is a solid-state method known as direct reduction in which briquettes or nuggets of iron ore are either mixed with coal or exposed to a reducing gas (such as natural gas or hydrogen) at elevated temperature. The direct reduction process does not involve iron ore agglomeration and cokemaking, processes which are significant energy users in the integrated steelmaking process. The carbon and hydrogen in these reducing agents reacts with the iron oxide to produce approximately 96% pure solid metallic iron, often referred to as sponge iron, in the form of nuggets or briquettes primarily for use in EAF steelmaking. In 2010, 75,000 metric tons were produced by the coal-based ITmk3® process at Mesabi Nugget, LLC in their startup year (capacity 500,000 metric tons (tonnes) per year) (Steel Dynamics. Inc. n.d.). Gas-based direct reduction plants are planned or under construction in the U.S. (including a Nucor plant in Louisiana and a Voestalpine plant in Texas) and this will affect the product mix in future years (Griggs 2013; Voestalpine 2013).

#### 2.2.4. Steelmaking

Steelmaking is the process where iron and scrap is converted to steel through the removal of impurities and incorporation of alloying elements.

#### 2.2.4.1. Basic Oxygen Furnace Steelmaking

After the molten pig iron is produced in the blast furnace, it is fed to a basic oxygen furnace along with a maximum of 30% scrap steel to be converted to raw steel. Oxygen is injected into the BOF in order to remove carbon from the pig iron through oxidation. The sensible heat in the molten pig iron and the heat generated by burning the carbon content with oxygen provide the energy required to melt the scrap and bring the bath to the required temperature for casting. The bath temperature is adjusted either by adding small amounts of carbon and then lancing with oxygen to increase the temperature or by adding iron ore to cool the bath. In 2010, 38.7% of steel produced in the U.S. was produced using the BOF route (AISI 2011a). However, a majority (70%) of world steel is produced through the BOF method (WSA 2012).

In BOF steelmaking, energy is lost through radiation and conduction from the vessel, unrecovered heat and calorific value of the off-gases, and through yield loss (oxidized iron in the slag, steel mixed into the slag by-product, oxidized particulates in the off-gas).

#### 2.2.4.2. Electric Arc Furnace Steelmaking

Raw steel produced via the electric arc furnace process is made from a feedstock of recycled steel and other iron scrap types, which makes this process distinctively different from BOF steelmaking. The process does not require a pig iron supply, as the energy to melt the charge is electrical and not dependent on high levels of carbon. An electric arc is produced between the feedstock and graphite electrodes to melt the recycled steel. Fuel (e.g., carbon fines plus oxygen via a side lance) is frequently used to accelerate the melting process. EAF steelmaking does not

involve iron ore agglomeration, cokemaking and ironmaking, processes which are significant energy users in the integrated steelmaking. In 2010, 61.3% of steel produced in the U.S. was produced using the EAF route (AISI 2011a). The share of EAF steelmaking has steadily increased since 1980, when 28% of steel was produced through this method (ArcelorMittal 2014).

In EAF steelmaking energy is lost through radiation and conduction, cooling of the furnace shell, unrecovered heat in the off-gases, and through yield loss (oxidized iron in the slag, steel mixed into the slag by-product, oxidized particulates in the off-gas, and oxidation through exposure of the liquid metal to air during tapping).

## 2.2.5. Casting

From the EAF or BOF, molten steel is semi-finished through casting into various solid forms, either individual ingots or continuous slabs, beams, billets, or blooms. Nearly all steel is cast using the continuous casting process, but a small percentage is still cast by the original batch method of pouring the molten steel into molds to create ingots. In 2010, 97.4% of steel produced in the U.S. was solidified using continuous casting while the remainder (2.6%) was solidified through the ingot method (AISI 2011a).

When transporting steel from the melting furnaces to the caster in a ladle, steelmakers can make minor adjustments to the composition and temperature, stir the metal with gas to improve its uniformity, and promote the flotation of undesirable contaminants such as slag and refractory material. More sophisticated adjustments in cleanliness, temperature, composition, and removal of dissolved gases are possible at an optional ladle metallurgy station. The flow pattern of liquid steel from the ladle to the caster (for example, through a bathtub-like tundish), and within the caster itself, is controlled to promote removal of detrimental inclusions. During the casting process, heat is removed from the steel as it cools from liquid to solid form-both to the surrounding environment and to chill water in the water-cooled continuous casting process. The cooling rates and thermal gradients are controlled to prevent cracking, control chemical segregation, and control the metallurgical structure of the steel. There is some yield loss, and therefore energy loss, due to removal of unusable portions of individual ingots. The yield loss is much smaller in continuous casting because most of the product can be used, apart from remnants of steel from the ladle and tundish ("skulls"). Improved yields have led to improved energy efficiency over time, most notably due to increased use of continuous casting vs. ingot casting.

## 2.2.6. Rolling

Rolling and finishing processes are used to produce the finished steel products. Steel rolling occurs in both hot and cold rolling mills. For hot rolling, the steel slabs or billets that are formed during continuous casting are reheated in furnaces to reach the necessary temperature and temperature uniformity before they are rolled to reduce their thickness. Although steel slabs or

billets are sometimes directly charged from continuous casting to a reheat furnace for hot rolling, those that are cooled to air temperature require longer reheat times and this increases the energy consumption of the process. Heat treatments and controlled processing may sometimes be performed during the hot rolling operation, for example, water quenching in a hot strip mill or a controlled temperature rolling in a plate mill. Electrical energy is required for the motors and hydraulics during the rolling process, steam is used to remove scale, and combustion gases are used for heating in the reheat furnaces.

Hot rolled coils can be sent to a cold rolling mill, where the thickness is further reduced—often with multiple steps and with intermediate high temperature heat treatments when the steel gets too hard to roll further without cracking. Final heat treatments are often required—for example, annealing to soften the steel for forming applications, or a quench and temper to increase the strength of the steel.

Yield loss, and therefore energy loss, can occur through oxidation of the steel, cracking of the edges of the rolled product during both hot rolling and cold rolling, and through cropping of the ends of the bars during hot rolling.

## 2.3. U.S. IRON AND STEEL MANUFACTURING ENERGY CONSUMPTION

Onsite energy and primary energy for the U.S. iron and steel sector are provided in Table 2-2. EIA MECS provides onsite energy consumption data by end use, including onsite fuel and electricity consumption, as well as feedstock energy. Primary energy includes assumptions for offsite losses (DOE 2014).

Table 2-2. U.S. Iron and Steel Manufacturing Energy Consumption Sector-Wide, 2010		
Onsite Energy Consumption (includes electricity, steam, and fuel energy used onsite at the facility)	999 TBtu	
Primary Energy Consumption (includes onsite energy consumption, and offsite energy losses associated with generating electricity and steam offsite and delivering to the facility)	1,359 TBtu	

Source: DOE 2014

Iron and steel manufacturing is the 5<sup>th</sup> largest consumer of energy in U.S. manufacturing, accounting for 1,359 TBtu (7%) of the 19,237 TBtu of total primary manufacturing energy consumption in 2010 (DOE 2014). Offsite electricity and steam generation and transmission losses in iron and steel manufacturing totaled 347 TBtu in 2010; onsite energy consumed within the boundaries of U.S. iron and steel mills totaled 999 TBtu.

Figure 2-3 shows the total onsite energy *entering* U.S. iron and steel mills; most of the energy entering is in the form of fuel. About 26% of this fuel is used onsite in boilers and combined heat and power (CHP) to generate additional electricity and steam (DOE 2014). In contrast, Figure

2-4 shows the total onsite energy at the *point of end use*. Electricity and steam from both offsite and onsite generation are included in Figure 2-4, along with the portion of energy loss that occurs in onsite generation. The data provided in Table 2-2, Figure 2-3, and Figure 2-4 are based on MECS with adjustments to account for withheld and unreported data (DOE 2014).



Figure 2-3. Onsite Energy Entering U.S. Iron and Steel Mills, 2010 (DOE 2014)



Figure 2-4. Onsite Energy Consumption at Point of End Use in U.S. Iron and Steel Mills, 2010 (DOE 2014)

#### 2.3.1. Fuel and Feedstocks

As shown in Figure 2-3, onsite fuel consumption amounted to 818 TBtu in 2010, or 83% of total onsite energy entering iron and steel mills (EIA 2013). Natural gas and blast furnace/coke oven gases account for over half (59%) of the sector's fuel consumption.

Figure 2-5 provides a breakdown of fuel consumption in the iron and steel sector by end use in 2010. The categories of end use are reported by EIA in MECS. The majority of the fuel (68%) is used directly for process heating; some examples of process heating equipment include fired heaters, heated reactors, and heat exchangers. A significant portion of fuel (26%) is used indirectly in boilers and CHP to generate additional onsite electricity and steam (DOE 2014).





Feedstock energy is the nonfuel use of combustible energy. For iron and steel manufacturing, feedstock energy is converted to products instead of being used as a fuel. For the energy bandwidth study, only the fuel use of combustible energy is considered in the opportunity analysis; however, due to the highly connected nature of feedstock and fuel energy, it is important to provide some context around that relationship and some background information on feedstock energy in this sector.

Figure 2-6 shows feedstock energy use in the U.S. iron and steel manufacturing sector. The sector consumed 373 TBtu of feedstock energy in 2010. Coal accounts for 87% of the feedstock energy used (EIA 2013).

It should be noted that a significant proportion of energy first used as feedstock is subsequently used as byproduct process energy. Comparison of MECS Tables 1.2, 2.2, and 3.2 indicates that 307 TBtu (82%) of feedstock energy is converted to byproduct onsite energy. MECS Table 3.5 indicates that 264 TBtu of byproduct onsite energy is blast furnace gas (BFG) and coke oven gas (COG), 2 TBtu is coke, and the remainder of byproduct energy is unclear.

(373 TBtu feedstock energy = 264 TBtu BFG/COG + 2 TBtu coke + remainder unclear)



Feedstock energy is a significant portion of energy consumption in U.S. iron and steel manufacturing (373 TBtu) so it is important to consider when analyzing overall iron and steel sector energy use. However, **feedstock energy is not included in the onsite energy data in the energy consumption bands in this study**. Feedstock energy is excluded in order to be consistent with previous bandwidth studies and because the relative amount of feedstock energy versus fuel energy used in manufacturing is not readily available for individual processes.

#### 2.3.2. Electricity

Figure 2-3 shows that onsite net electricity entering iron and steel mills totaled 172 TBtu in 2010. The data presented is the *net amount*, which is the sum of purchases and transfers from offsite sources as well as generation from non-combustion renewable resources (e.g., hydroelectric, geothermal, solar, or wind energy) less the amount of electricity that is sold or transferred out of the plant. Figure 2-4 shows that 188 TBtu of total electricity is consumed at the point of end use and includes 16 TBtu of electricity generated onsite.

In Figure 2-7, the breakdown of the 188 TBtu of electricity is shown by end use in 2010 (DOE 2014). There are numerous uses for electricity in iron and steel manufacturing; the most common uses are for process heating and machine driven equipment (i.e., motor-driven systems such as compressors, fans, pumps, and materials handling and processing equipment). Motors used for cooling water circulation pumps and fans, however, are accounted for in process cooling end use. Other end uses of electricity for iron and steel manufacturing are less significant, but include nonprocess facility related end uses (e.g., facility heating, ventilation, and air conditioning (HVAC), facility lighting, cooking, office equipment, etc.) and other end uses.





#### 2.3.3. Steam

Figure 2-3 shows 8 TBtu of net steam entering iron and steel mills in 2010. The data presented is the *net amount*, which is the sum of purchases, generation from renewables, and net transfers. A larger amount of steam is generated onsite. Figure 2-4 shows that 161 TBtu of steam is consumed at the point of end use, including 152 TBtu of steam generated onsite (34 TBtu of purchased and generated steam is lost through distribution to end uses) (DOE 2014).

Figure 2-8 shows the breakdown of 127 TBtu of steam by end use in 2010 (DOE 2014). A majority of the offsite- and onsite-generated steam is used for process heating; other end uses for steam in iron and steel manufacturing include facility heating, ventilation, and air conditioning (HVAC), other process uses, machine driven equipment (i.e., steam turbines), , and other nonprocess end uses. Unlike fuel and electricity end use, steam end use is <u>not</u> reported in MECS. The end use distribution shown here was determined in the Energy Footprint analysis (DOE 2014) based on input from an industry-led working group.



Figure 2-8. Steam Consumption in the Iron and Steel Sector by End Use, 2010 (DOE 2014)

20 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing
# **3.** Production Volumes in U.S. Iron and Steel Manufacturing

In this bandwidth study, six iron and steel processes were selected for individual analysis. Table 3-1 presents the U.S. production data of the six processes for the year 2010, unless otherwise noted.

Table 3-1. U.S. Iron and Steel Production Volumes for Processes           Studied in 2010			
Process	Production (1,000 tons)		
Agglomeration Pelletizing <sup>a</sup> Sintering <sup>a</sup>	39,728 5,759		
Cokemaking <sup>b</sup>	9,292 (integrated only) 12,021 (integrated + offsite)		
Ironmaking <sup>c</sup>	29,590		
Steelmaking			
BOF	34,345		
EAF	54,386		
Casting (Continuous)	84,784		
Rolled Products			
Heavy Shapes and Plates	14,899		
Tin Plate and Products	2,791		
Bar	14,128		
Rod	4,213		
Sheets and Strip	45,783		

<sup>a</sup> Represents pellet and sinter consumption in steel mill blast furnaces; assumed to be produced domestically.

<sup>b</sup> The smaller value represents production of coke at integrated steel mills; the larger value represents total coke consumption including coke produced outside of integrated steel mills, e.g., merchant coke.

<sup>c</sup> Excludes limited production of ITmk3-produced iron nuggets.

Source: AISI 2011a.

The most energy intensive processes were selected for this study. Other, less intensive processes were added to the study to expand the representative coverage of the iron and steel sector as a whole. In general, the selection of processes was largely dependent on the availability of current production and energy consumption data.

The year 2010 was used for production values to correspond with the latest MECS data, which is also for 2010. Production data was gathered from the American Iron and Steel Institute (AISI), the leading source for information on steel production in North America. The AISI Statistical Summary is released annually and provides production data along with other statistical information.

22 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

# 4. Current Typical Energy Consumption for U.S. Iron and Steel Manufacturing

This Chapter presents the energy consumption data for individual iron and steel manufacturing processes and sector-wide in 2010. Energy consumption in a manufacturing process can vary for diverse reasons. The energy intensity estimates reported herein are representative of average U.S. iron and steel manufacturing; they do not represent energy consumption in any specific facility or any particular region in the United States.

# 4.1. BOUNDARIES OF THE IRON AND STEEL BANDWIDTH STUDY

Estimating energy requirements for an industrial process depends on the boundary assumptions; this is especially true in the iron and steel industry. The key focus of this bandwidth study is energy consumption within the plant boundary, which is the *onsite* use of process energy (including purchased energy and onsite generated steam and electricity) that is directly applied to iron and steel manufacturing.

This study does not consider lifecycle energy consumed during raw material extraction, off-site treatment, and transportation of materials. Upstream energy, such as the energy required for processing and handling materials outside of the plant is also not included. To be consistent with previous bandwidth studies, feedstock energy and the energy associated with delivering feedstocks to the plant gate (e.g., producing, conditioning, and transporting feedstocks) are *excluded* from the energy consumption bands in this analysis.

# 4.2. ESTIMATED ENERGY INTENSITY FOR INDIVIDUAL PROCESSES

Energy intensity data are needed to calculate bands of energy consumption in this study. This Section presents the estimated energy intensities of the six processes studied.

The specific energy needed to make a ton of product can vary significantly between processes, and also between facilities. Energy intensity is a common measure of energy performance in manufacturing. Energy intensity is reported in units of energy consumption (typically Btu) per unit of manufactured product (typically short tons, tons, or metric tons) and, therefore, reported as million Btu per short ton (MMBtu/ton). Energy intensity estimates are available for specific equipment performance, process unit performance, or even plant-wide performance. Energy intensity can be estimated by process, both in the United States and other global regions, based on average, representative process and plant performance.

Appendix A1 presents the CT energy intensities and energy consumption for the six processes studied. Table 4-1 presents a summary of the references consulted to identify CT energy intensity by process. Appendix A2 provides the references used for each process. Appendix A3 provides a breakdown of CT energy intensity by energy type. In cases where energy intensity is reported in terms of lower heating value LHV (or net calorific), a ratio of higher heating value HHV to LHV

was used to convert to HHV. HHV conversions by fuel type for CT and SOA energy intensity are provided in Appendix A3.

Because the iron and steel sector is diverse, covering many products, a range of data sources were considered (see Table 4-1). In most cases, multiple references were considered for each process.

Many sources were consulted in order to determine the energy intensity of the six processes selected for study. Each iron and steel facility is unique and steel is produced in different scales and by different processes; thus, it is difficult to ascertain an exact amount of energy necessary to produce a certain volume of a product. For instance, ore quality differs in chemical composition and iron content which affects the amount of energy needed for the reduction reaction to produce iron. Plant size can also impact operating practices and energy efficiency. Higher efficiency is often easier to achieve in larger plants. Consequently, the values for energy intensity provided should be regarded as estimates based on the best available information.

Studied	
Source	Description
AISI 2011a	Summary for steel industry statistics, published by the American Iron and Steel Institute. The report for year 2010 is referenced.
EIA 2013a	Manufacturing Energy Consumption Survey data released by EIA every four years; this data comes from a survey that is taken by U.S. manufacturers. The most recent year for which MECS data is published is 2010. The data is scaled up to cover the entirety of U.S. manufacturing and for individual manufacturing subsectors.
EIA 2013b	Includes documentation for the model EIA utilizes to project industrial energy use.
Energetics 2000	The Energy and Environmental Profile of the U.S. Iron and Steel Industry, prepared by Energetics and published by DOE in 2000 provides a detailed breakdown (including total processing energy) for key process areas.
EPA 2012	This 2012 report by the Environmental Protection Agency (EPA) provides a list of energy efficiency improvement measures for use by the iron and steel industry.
IEA 2007	This 2007 report by the International Energy Agency (IEA) includes a chapter focused on iron and steel.
IPPC 2013	While this bandwidth analysis focuses on the U.S.; specific European energy consumption values or ranges are listed for select processes in this report.
NRC 2007	Provides graphics showing actual consumption at multiple Canadian plants; also addresses best available technologies.
Stubbles 2000	This report details energy consumption in the U.S. steel industry for various processes.

Table 4-1. Published Sources Reviewed to Identify Current Typical Energy Intensities for Processes
Studied

When available, energy intensity values specific to the United States were used in calculating CT energy consumption for the six select processes. For certain processes, the energy intensity data was only available for non-U.S. regions, such as Canada or Western Europe. This data was used with the understanding that energy consumption can vary from country to country and from region to region. Energy intensity for the six process areas studied is summarized in Appendix A1 and detailed by process area and energy type in Appendix A3. In cases where energy intensity is reported in terms of lower heating value LHV (or net calorific), a ratio of higher heating value HHV to LHV is used to convert to HHV. HHV conversions by fuel type for CT and SOA energy intensity are provided in Tables A3-27 and A3-28.

## 4.2.1. Ore Agglomeration

Pelletizing energy intensity is dependent on the type and quality of iron ore and whether limestone is used to produce fluxed pellets. Pelletizing is conducted primarily at the mining site in the U.S. (NAICS 212210 Iron Ore Mining), and thus energy requirements for pelletizing are not included in U.S. government steel industry energy data. Limited data is available regarding current energy use in pelletizing. The European Union, in their Best Available Techniques reference document, reports a value of approximately 0.70 MMBtu/ton (0.82 GJ/tonne) for pelletizing using hematite ore (IPPC 2013). It should be noted that energy requirements may not be comparable to U.S. values due to differences in ores; or reflect other ore processing requirements. Also, magnetite pelletizing energy consumption would be lower, as magnetite oxidation provides a large portion of the energy required. As previously indicated, pelletizing energy is not included in the actual bandwidth calculations but this value is provided for reference.

The sintering energy intensity estimate also is derived from IPPC 2013. The estimated energy intensity of 1.32 MMBtu/ton sinter (1.54 GJ/tonne) is somewhat lower than previous values reported by others such as the International Energy Agency, which had reported a range of 1.29-1.72 MMBtu/ton sinter (1.50-2.00 GJ/tonne) (IEA 2007). However, overall energy consumption for sinter production at U.S. plants based on this value totals less than 10 TBtu, less than 1% of steel industry energy use, assuming all sinter consumption is domestically produced.

## 4.2.2. Cokemaking

Cokemaking energy intensity is derived from both values provided by AISI in their annual statistical report (AISI 2011a) and estimates from EIA 2013b. It should be noted that U.S. cokemaking energy values, estimated to be 3.83 MMBtu/ton coke (4.45 GJ/tonne), are specific to cokemaking facilities at integrated steel plants. In addition to process energy requirements, a moderate amount of energy ends up in cokemaking chemical by-products. Merchant coke plants, which on the whole are newer, are not accounted for in the AISI statistical report and fall under the petroleum and coal products industry within NAICS (324199 All Other Petroleum and Coal Products Manufacturing). The estimated value is also comparable to individual plant values listed in NRC 2007.

#### 4.2.3. Ironmaking

Blast furnace ironmaking energy intensity is also derived from AISI statistical report data along with other estimates from the IPPC report (AISI 2011a, IPPC 2013). Blast furnace ironmaking energy intensity is estimated at 11.72 MMBtu/ton of hot metal (13.63 GJ/tonne). With many iron and steel plants operating at less than full capacity in 2010, this value may be slightly higher than normal.

While the traditional direct reduction route was not used in the U.S. in 2010 and data was not available for the limited production via the ITmk3® process (thus no calculated current typical), the IEA has reported that both of the gas-based direct reduction processes use around 8.9 MMBtu/ton (10.4 GJ/tonne) natural gas; with some additional electricity use required (80-90 kWh/tonne) which could be cogenerated onsite (IEA 2007).

#### 4.2.4. Steelmaking

Steelmaking in the basic oxygen furnace (BOF) with subsequent refining operations (ladle metallurgy, vacuum treatment, slag management, etc.) can have a fairly wide range of energy use. Energy intensity for BOF steelmaking is estimated at 0.58 MMBtu/ton of liquid steel (0.67 GJ/tonne) (IPPC 2013). Of note in 2010, it was estimated that approximately 24% of U.S. BOF inputs came from scrap and not pig iron (USGS 2012c). Also, Stubbles noted that while recovery of chemical energy in the off-gas from BOF operations is possible, the economics of doing so in the U.S. were not attractive at the time of his report (Stubbles 2000). A small amount of carbon feedstock remains in the liquid steel.

Steelmaking in electric arc furnaces (EAFs) also can have a fairly wide range of energy use, and also differences in fuel mix. Overall, scrap accounts for over 90% of metal inputs, with lesser amounts of pig iron and imported direct-reduced iron (DRI) accounting for the remainder. In 2010, it was estimated that approximately 91% of U.S. EAF inputs came from scrap (USGS 2012c). While steelmakers are increasing the share of chemical energy, electricity still generally accounts for the largest share of energy consumption in the EAF. Thus for individual furnaces, a range of both chemical energy and total energy is found; and many of the smaller furnaces are significantly less efficient. For 2010, final energy intensity for EAF steelmaking is estimated at 1.86 MMBtu/ton liquid steel (2.16 GJ/tonne) using data from the Association for Iron and Steel Technology (AIST 2011). Stubbles (2000) noted that an estimated additional 100 kilowatt hours (kWh) are used for ladle heating, cranes, and other auxiliary services at EAF plants; these electric-consuming operations may not be included in the AIST-based estimates.

#### 4.2.5. Casting and Rolling

In 2010, over 97% of raw steel in the U.S. was continuously cast, which consumes about 0.19 MMBtu/ton (0.22 GJ/tonne), based on the average of values from NRC 2007. Traditional hot rolling operations consume an estimated average of 2.58 MMBtu/ton (3.00 GJ/tonne) – this is considering both hot strip and plate as well as section mills from NRC 2007. Cold rolling mill

operations consume an estimated 3.48 MMBtu/ton (4.05 GJ/tonne) (EIA 2013b), though this number will differ whether rolling strip, bar, or wire. Tinplate and galvanized steel would have higher energy requirements, but values are not estimated in this report.

# 4.3. CALCULATED CURRENT TYPICAL ENERGY CONSUMPTION FOR INDIVIDUAL PROCESSES

Table 4-2 presents the calculated onsite CT energy consumption for the six processes studied. To calculate onsite CT energy consumption, energy intensity for each process (presented initially in Appendix A1) is multiplied by the 2010 production data (presented initially in Table 3-1 and also in Appendix A1). Feedstock energy is excluded from the consumption values. The CT energy consumption for these six processes is estimated to account for 822 TBtu of onsite energy, or 82% of the 999 TBtu of sector-wide onsite energy use in 2010. Appendix A1 also presents the onsite CT energy consumption for the six processes individually.

Calculated primary CT energy consumption by process is also reported in Table 4-2. Primary energy includes offsite energy generation and transmission losses associated with electricity and steam from offsite sources. To determine primary energy, the net electricity and net steam portions of sector-wide onsite energy are scaled to account for offsite losses and added to onsite energy (see the footnote in Table 4-2 for details on the scaling method).

Table 4-2. Calculated U.S. Onsite Current Typical Energy Consumption for Processes Studied in 2010 with Calculated Primary Energy Consumption and Offsite Losses

Process	Energy Intensity (MMBtu/ton) [GJ/tonne]	Production (1,000 ton/year)	Onsite CT Energy Consumption, Calculated (TBtu/year)	Offsite Losses, Calculated (TBtu/year) <sup>2</sup>	Primary CT Energy Consumption, Calculated (TBtu/year)
Integrated Mills					
Agglomeration Pelletizing <sup>b</sup> Sintering	0.70 [0.82] 1.32 [1.54]	39,728 5,759	- 8	- 1	- 9
Cokemaking <sup>c</sup>	3.83 [4.46]	9,292	36	2	37
Ironmaking Blast Furnace <sup>d</sup>	11.72 [13.63]	29,590	337	14	351
Steelmaking BOF <sup>e</sup>	0.58 [0.67]	34,345	20	8	28
Mini Mills					
Ironmaking Direct Reduction <sup>f</sup>	9.17 [10.71]	negligible in 2010	-	-	-
Steelmaking EAF <sup>e</sup>	1.86 [2.16]	54,386	101	155	256
Integrated and Mini Mills					
Casting	0.19 [0.22]	84,784	16	12	28
Rolling Hot Cold <sup>d</sup>	2.58 [3.00] 3.48 [4.05]	84,784 27,710	219 86	63 59	282 145
Total for Processes Studied			822		

Current Typical (CT)

<sup>1</sup> Accounts for offsite electricity and steam generation and transmission losses. Offsite electrical losses are based on published grid efficiency. EIA Monthly Energy Review, Table 2.4, lists electrical system losses relative to electrical retail sales. The energy value of electricity from offsite sources including generation and transmission losses is determined to be 10,553 Btu/kWh. Offsite steam generation losses are estimated to be 20% (Swagelok Energy Advisors, Inc. 2011. <u>Steam Systems Best Practices</u>) and offsite steam transmission losses are estimated to be 10% (DOE 2007, <u>Technical Guidelines Voluntary Reporting of Greenhouse Gases</u> and EPA 2011, <u>ENERGY STAR Performance Ratings Methodology</u>).

References for production data and energy intensity data are provided by process in Appendix A2. The other values are calculated as explained in the text.

# 4.4. CURRENT TYPICAL ENERGY CONSUMPTION BY PROCESS AND SECTOR-WIDE

In this Section, the CT energy consumption estimates for nine processes studied are provided.

Table 4-3 presents the onsite CT energy consumption by process and sector-wide for U.S. iron and steel manufacturing. The six processes studied account for 82% of all onsite energy consumption by the U.S. iron and steel sector in 2010. As shown in the last column of Table 4-3, the percentage of coverage of the processes studied is calculated. This indicates how well the processes studied represent total sector-wide MECS-reported energy. The overall percentage of coverage for the processes studied (82%) is used later in this study to determine the extrapolated total sector-wide SOA, PM, and TM energy consumptions.

Table 4-3 also presents CT primary energy consumption by process. Primary energy is calculated from onsite CT energy consumption databased on an analysis of MECS data (DOE 2014), with scaling to include offsite electricity and steam generation and transmission losses (DOE 2014).

Table 4-3. Onsite and Primary Current Typical Energy Consumption for the Nine Processes Studied
and Sector-Wide in 2010, with Percent of Sector Coverage

Process	Onsite CT Energy Consumption, calculated (TBtu/year)	Primary CT Energy Consumption, calculated* (TBtu/year)	Percent Coverage (Onsite CT as a % of Sector-wide Total)**
Integrated Mills			
Agglomeration Pelletizing <sup>a</sup> Sintering	- 8	- 9	- 1%
Cokemaking <sup>b</sup>	36	37	4%
Ironmaking Blast Furnace <sup>c</sup>	337	351	34%
Steelmaking BOF <sup>d</sup>	20	28	2%
Mini Mills			
Ironmaking Direct Reduction <sup>e</sup>	-	-	-
Steelmaking EAF <sup>d</sup>	101	256	10%
Integrated and Mini Mills			
Casting	16	28	2%
Rolling Hot Cold <sup>°</sup>	219 86	282 145	22% 9%
<b>Total for Processes Studied</b>	822	1,135	82%
All Other Processes	176	213	18%
Total for Iron and Steel Sector- wide	999***	1,348***	100%

Current typical (CT)

\* Accounts for offsite electricity and steam generation and transmission losses. Offsite electrical losses are based on published grid efficiency. EIA Monthly Energy Review, Table 2.4, lists electrical system losses relative to electrical retail sales. The energy value of electricity from offsite sources including generation and transmission losses is determined to be 10,553 Btu/kWh. Offsite steam generation losses are estimated to be 20% (Swagelok Energy Advisors, Inc. 2011. <u>Steam Systems</u> <u>Best Practices</u>) and offsite steam transmission losses are estimated to be 10% (DOE 2007, <u>Technical Guidelines Voluntary</u> <u>Reporting of Greenhouse Gases</u> and EPA 2011, <u>ENERGY STAR Performance Ratings Methodology</u>).

\*\* Calculated by dividing the onsite CT energy consumption for the processes studied by sector-wide onsite CT energy consumption (999 TBtu).

\*\*\* Source for sector-wide values is DOE 2014.

<sup>a</sup> The pelletizing estimate is for a single facility, Pelletizing is not included in iron and steel mill energy consumption totals as this process takes place offsite.

<sup>b</sup> Cokemaking onsite and primary energy excludes approximately 300 TBtu of nonfuel feedstock coke.

<sup>c</sup> Blast furnace and cold rolling onsite and primary energy consumption each exclude 10 TBtu of nonfuel feedstock natural gas. <sup>d</sup> In 2010, 24% of BOF inputs and 91% of EAF inputs came from scrap.

<sup>e</sup> As there was limited DRI production in the U.S. in 2010, the CT energy consumption values are shown for illustration only.

# 5. State of the Art Energy Consumption for U.S. Iron and Steel Manufacturing

As plants age, manufacturing processes and equipment are updated and replaced by newer, more energy-efficient technologies. This results in a range of energy intensities among U.S. iron and steel mills. Iron and steel mills will vary widely in size, age, efficiency, energy consumption, and types and amounts of products. Modern iron and steel mills can benefit from more energy-efficient technologies and practices.

This Chapter estimates the energy savings possible if U.S. iron and steel mills adopt the best technologies and practices available worldwide. State of the art (SOA) energy consumption is the minimum amount of energy that could be used in a specific process using existing technologies and practices.

# 5.1. CALCULATED STATE OF THE ART ENERGY CONSUMPTION FOR INDIVIDUAL PROCESSES

Appendix A1 presents the onsite SOA energy intensity and consumption for the six processes considered in this bandwidth study in alphabetical order. The SOA energy consumption for each iron and steel manufacturing process is calculated by multiplying the SOA energy intensity for each process by the relevant production (all relevant data are presented in Appendix A1).

The onsite SOA energy consumption values are the net energy consumed in the process using the single most efficient process and production pathway. No weighting is given to processes that minimize waste, feedstock streams, and byproducts, or maximize yield, even though these types of process improvements can help minimize the energy used to produce a pound of product. The onsite SOA energy consumption estimates exclude feedstock energy.

Table 5-1 presents the published sources referenced to identify the SOA energy intensities.

 Table 5-1. Published Sources Referenced to Identify State of the Art Energy Intensities for Nine Select

 Processes

Source Abbreviation	Description
Energiron 2013	Energiron: The Innovative Direct Reduction Technology, Information on the direct reduction process
Giavani et al. 2012	Consteel Evolution <sup>1M</sup> – The Second Generation of Consteel Technology, Information regarding a specific EAF technology, the Consteel Evolution
IPPC 2013	Best Available Techniques (BAT) Reference Document for Iron and Steel Production, European Commission. Integrated Pollution Prevention and Control. While this bandwidth analysis focuses on the U.S., this report lists specific European energy consumption values or ranges for select processes
LBNL 2008	This Lawrence Berkeley National Laboratory report, <i>World Best Practice Energy</i> <i>Intensity Values for Selected Industrial Sector</i> , provides best practice values for many industrial processes, including iron and steelmaking.
NRC 2007	Benchmarking Energy Intensity in the Canadian Steel Industry, Natural Resources Canada. This report provides graphics and data for a variety of processes using best available technologies (energy-saving technologies that are both commercially available and economically attractive); the report also provides actual consumption at multiple Canadian plants.

NRC 2007 is heavily referenced in determining SOA intensity estimates. Technologies employed are a deviant of the International Iron and Steel Institute (IISI) EcoTech Plant which includes energy-saving technologies that are both commercially available and economically attractive, with additional inclusion of certain technologies that, while less economically attractive, were being utilized commercially in Canadian steel plants.

Technologies identified that are in a pre-commercial stage of development or that are extremely expensive were not considered in the SOA analysis (instead they were considered in Chapter 6 on the practical minimum (PM) energy consumption).

#### 5.1.1. Ore Agglomeration

NRC 2007 does not include sintering. For sinter plants, the state of the art value of 1.27 MMBtu/ton (1.47 GJ/tonne) of sinter is derived from LBNL 2008 and IPPC 2013 has numbers that are very comparable.

#### 5.1.2. Cokemaking

The value of 3.37 MMBtu/ton (3.92 GJ/tonne) of coke from NRC 2007 employs technologies including coke dry quenching, enhanced combustion control, and high-pressure ammonia liquor spray aspiration, among others.

#### 5.1.3. Ironmaking

For integrated plants utilizing blast furnaces, the value of 11.13 MMBtu/ton of hot metal (12.94 GJ/tonne) from NRC 2007 includes technologies for hot stoves including waste gas heat recovery, oxygen enrichment of cold blast, high efficiency stoves, and combustion control, and top gas recovery turbine for the blast furnace.

While direct iron reduction (DRI) was not employed extensively in the U.S. in 2010, this will be changing in 2013 with the expected opening of a Nucor plant currently under construction in Louisiana. Gas-based DRI processes are generally the most efficient, with Midrex and Energiron (joint venture between Tenova, Danieli, and HYL) shaft furnaces accounting for the majority of gas-based plants. Data from Energiron indicate best values of 8.3 MMBtu/ton (9.7 GJ/tonne). CHP is an option to provide electricity demand at these plants as well.

#### 5.1.4. Steelmaking

For integrated plants the value of -0.30 MMBtu/ton (-0.35 GJ/tonne) from NRC 2007 includes the basic oxygen furnace and secondary refining: gas recovery with dry gas cleaning system, steam recovery, ladle management, and bottom stirring.

For EAF steelmaking, two values are provided; one from NRC 2007 which utilizes more electricity; and also Consteel Evolution which has greater amount of fuel use and extensive scrap or charge preheating. Oxygen production is also assumed to utilize state of the art technology. Depending on whether one is evaluating final or primary energy as measure of efficiency will lead to different technology use. For NRC 2007, final energy intensity is calculated at 1.62 MMBtu/ton (1.89 GJ/tonne). For Consteel Evolution, with a lower share of electricity use, final energy intensity is calculated at 1.79 MMBtu/ton (2.08 GJ/tonne) but has a lower primary energy intensity. The value from NRC 2007 is used for calculations for consistency while the Consteel Evolution value is provided for comparison purposes.

#### 5.1.5. Casting and Rolling

For continuous casting, the state of the art value is estimated at 0.05 MMBtu/ton (0.06 GJ/tonne) (NRC 2007). An estimate for thin slab casting and rolling energy intensity is also provided in Appendix A3, although not referenced in the energy consumption analysis; LBNL 2008 estimates the best value for this slab casting to be MMBtu/ton (0.20 GJ/tonne).

For a traditional hot strip mill, the best value is estimated at 1.43 MMBtu/ton (1.66 GJ/tonne) (NRC 2007), with variations for section mills (whether the product is rod, bar, or heavy) as depicted in Appendix Table A3-17. The average value of 1.60 MMBtu/ton (1.86 GJ/tonne) is used for hot rolling overall. Technologies employed to achieve the hot rolling values include recuperation and heat recovery, furnace controls, transfer bar edge heaters, energy efficient burners, and high efficiency motors.

For cold rolling mills, LBNL 2008 estimates best value for final energy of 1.42 MMBtu/ton (1.645 GJ/tonne) inclusive of pickling, cold rolling, and annealing.

#### 5.1.6. Other Estimates of SOA Savings

DOE AMO conducted energy saving assessments at 53 plants in NAICS code 3311 between 2006 and 2011. The assessments focused primarily on process heating and steam applications, with the vast majority of assessments occurring at EAF plants. On average, existing opportunities identified in these assessments were predicted to save 6.2% of current energy consumption (DOE 2013).

EPA published a report in September 2012 (EPA 2012) that describes available and emerging technologies for reducing greenhouse gas emissions in the iron and steel industry; AISI provided input on technical feasibility and cost-effectiveness for various technology options. Technical details of energy efficiency measures are provided for specific industry processes, along with information on technology readiness and estimated payback. Overall savings potential is not provided in this report.

Lawrence Berkeley National Laboratory also published a study (separate to the study referenced in Table 5-1) that identifies energy efficiency improvement and cost saving opportunities in the iron and steel sector. This ENERGY STAR guide provides information on a range of measures, including crosscutting energy management program savings, energy systems savings (steam, motors, pumps, fans, compressed air), and process specific savings. Individual savings and paybacks are provided for some measures, combined savings is not available.

# 5.2. STATE OF THE ART ENERGY CONSUMPTION BY PROCESS AND SECTOR-WIDE

Table 5-2 presents the onsite SOA energy consumption for the six U.S. iron and steel processes studied. In this table, the SOA energy consumptions for the processes studied are summed and extrapolated to provide a sector-wide onsite SOA energy consumption. Table 5-2 also presents the onsite SOA energy savings, or the *current opportunity*. The SOA energy savings is also expressed as a percent in Table 5-2. This is also shown in Figure 5-1. It is useful to consider both TBtu energy savings and energy savings percent when comparing the energy savings opportunity. Both are good measures of opportunity; however, the conclusions are not always the same. In Figure 5-1, the percent savings is the percent of the overall energy consumption bandwidth, with CT energy consumption as the upper benchmark and TM as the lower baseline. In Figure 5-2, the *current* energy savings opportunity is shown in terms of TBtu/year savings for each process. The pie chart in Figure 5-2 captures the blue portions of the bar chart shown in Figure 5-1. Among the processes studied, the greatest *current opportunity* in terms of percent energy savings is casting at 74% energy savings; the greatest *current opportunity* in terms of TBtu savings is hot rolling at 83 TBtu per year savings.

To extrapolate the sector-wide data presented in Table 5-2 and Figure 5-1, the SOA energy consumption of each individual process studied is summed, and the sum is divided by the percent coverage for the entire subsector (82%, as shown in Table 4-3). Percent coverage is the ratio of the sum of all the CT energy consumption for the individual processes studied to the CT energy consumption for the subsector provided by MECS (see Table 4-3). The extrapolated number is the estimated SOA energy consumption for the entire subsector. The SOA energy consumption for the iron and steel sector (i.e., all processes that are not included in the six processes studied) was calculated by subtracting the total for the processes studied from the sector-wide total. These additional processes are together referred to as All Other Processes in Table 5-2).

Table 5-2 also presents the SOA energy savings percent. To calculate the onsite SOA energy savings percent, the thermodynamic minimum (TM) energy consumption serves as the baseline for estimating percent energy savings, not zero. The energy savings percent is the percent of energy saved with SOA technologies and practices compared to CT energy consumption, considering that the TM may not be zero. As will be explained in Chapter 7, the TM reaction energy for one iron and steel manufacturing processes is a negative value. When comparing energy savings percent from one process to another, the absolute savings is the best measure of comparison. The equation for calculating onsite SOA energy savings percent is:

$$SOA Savings \% = \frac{CT - SOA}{CT - TM}$$

 Table 5-2. Onsite State of the Art Energy Consumption, Energy Savings, and Energy Savings Percent for the

 Processes Studied and Sector-Wide

Process	Onsite CT Energy Consumption (TBtu/year)	Onsite SOA Energy Consumption (TBtu/year)	SOA Energy Savings <sup>↑</sup> (CT-SOA) (TBtu/year)	SOA Energy Savings Percent (CT-SOA)/ (CT-TM)*
Integrated Mills				
Agglomeration Sintering	8	7	<1	17%
Cokemaking	36	31	4	22%
Ironmaking Blast Furnace	337	329	8	9%
Steelmaking BOF	20	-10	30	72%
Mini Mills				
Steelmaking EAF	101	88	13	33%
Integrated and Mini Mills				
Casting**	16	4	12	74%
Rolling** Hot Cold	219 86	136 39	83 47	38% 55%
Total for Processes Studied	822	625	197	39%
All Other Processes	176	134	42	39%
Total for Iron and Steel Sector-wide	999	759***	240	39%

Current typical (CT), State of the art (SOA)

*†* SOA energy savings is also called Current Opportunity.

\* SOA energy savings percent is the SOA energy savings opportunity from transforming iron and steel manufacturing processes. Energy savings percent is calculated using TM energy consumption shown in Table 7-2 as the minimum energy consumption. The energy savings percent, with TM as the minimum, is calculated as follows: (CT- SOA)/(CT- TM)

\*\* Does not include thin slab casting, which also replaces rolling; thin slab casting is not included in the subtotal or total values.

\*\*\* The sector-wide SOA energy consumption was an extrapolated value, calculated by dividing the total onsite SOA energy consumption for the processes studied by the overall percent coverage from Table 4-3 (82%).



Figure 5-1. Current Opportunity Energy Savings Bandwidths for Processes Studied (with Percent of Overall Energy Consumption Bandwidth)



#### Current Energy Savings Opportunity by Process

#### 5.2.1. Comparing State of the Art and Current Typical Energy Data

If all U.S. iron and steel mills were able to attain onsite SOA energy intensities, it is estimated that 197 TBtu per year of energy could be saved from the six processes alone, corresponding to a 39% energy savings sector-wide. This energy savings estimate is based on adopting available SOA technologies and practices without accounting for future gains in energy efficiency from R&D. This is a simple estimate for potential savings, it is not inferred that all existing mills could achieve these state of the art values or that the improvements would prove to be cost effective in all cases. The interactivity between savings measures is also not specifically considered. For instance, increased use of pulverized coal in the blast furnace would reduce coke requirements and associated energy use.

Noting that the energy values described in Table 5-2 are based on 2010 production values and processing routes, it is important to remember that these values are not static, but dynamic in nature. As previously discussed, steel mills operated at less than optimal capacity in 2010, negatively impacting energy efficiency to some extent as shown in an AISI chart depicting energy intensity over time as shown in Figure 5-3. In 2013, a more efficient DRI plant opened in the U.S. and the ITmk3® process is also now in commercial operation at one U.S. plant. As the supply of recycled steel (scrap) is ever-changing, the quantity of production required from virgin materials will continue to depend on market and economic conditions.





40 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

# 6. Practical Minimum Energy Consumption for U.S. Iron and Steel Manufacturing

Technology innovation is the driving force for economic growth. Across the globe, R&D is underway that can be used to make iron and steel products in new ways and improve energy and feedstock efficiency. Commercialization of these improvements will drive the competitiveness of U.S. iron and steel manufacturing. In this Chapter, the R&D energy savings made possible through R&D advancements in iron and steel manufacturing are estimated. Practical minimum (PM) is the minimum amount of energy required assuming the deployment of applied R&D technologies under development worldwide.

## 6.1. R&D IN THE IRON AND STEEL INDUSTRY

Investing in R&D in the short term ensures long term future prosperity. Increasing the energy efficiency of an existing process often requires capital investment and taking manufacturing equipment offline to perform the necessary updates. The risks and rewards of this type of business decision needs to be clearly assessed. As the iron and steel industry is relatively mature, most iron and steel companies (along with other primary metal producers) typically allocate less than 1% of their annual sales toward R&D (NSF 2007).

Extensive efforts have been undertaken to develop alternative approaches to making iron and steel in particular, with potential to reduce both energy intensity and carbon emissions. Various groups including the European ULCOS (Ultra–Low Carbon Dioxide Steelmaking) consortium, Hatch, and others have conducted efforts to model the energy requirements and carbon emissions from alternative ore-based process routes. One way processing route options have been depicted is within a triangle framework, with carbon, hydrogen, and electrons representing the three vertices, as can be seen in Figure 6-1. Various approaches are shown on the figure which would shift iron and steel production away from coal (decarbonizing), where carbon would be replaced by hydrogen or electricity in processes such as hydrogen reduction or electrolysis of iron ore.



# 6.2. CALCULATED PRACTICAL MINIMUM ENERGY CONSUMPTION FOR INDIVIDUAL PROCESSES

In this study, PM energy intensity is the estimated minimum amount of energy consumed in a specific iron and steel process assuming that the most advanced technologies under research or development around the globe are deployed.

R&D progress is difficult to predict and potential gains in energy efficiency can depend on financial investments and market priorities. To estimate PM energy consumption for this bandwidth analysis, a broad search of R&D activities in the iron and steel industry was conducted. A large number and range of potential technologies were identified.

The focus of this study's search was applied research, which was defined as investigating new technology with the intent of accomplishing a particular objective. Basic research, the search for unknown facts and principles without regard to commercial objectives, was not considered. Many of the technologies identified were disqualified from consideration due a lack of data from which to draw energy savings conclusions.

Appendix A1 presents the onsite PM energy consumption for the six processes considered in this bandwidth study. The PM energy consumption for each process is calculated by multiplying the estimated PM energy intensity for each process by the process's 2010 production volume (the energy intensity and production data are also presented in Appendix A1). These values exclude feedstock energy. The lower limit for onsite PM energy intensity and onsite PM energy consumption are presented in Appendix A1. The upper limit of the PM range is assumed to be the SOA energy consumption. The PM energy consumption for each process is expressed as a range because the energy savings impacts are speculative and based on unproven technologies.

Table 6-1 presents the key sources consulted to identify PM energy intensities in iron and steel manufacturing. Additionally, numerous fact sheets, case studies, reports, and award notifications were referenced; a more detailed listing of references is provided in Appendix A4 (Table A4 and References for Table A4).

Processes	
Reference Abbreviation	Source
Birat et al. 2009	"The "CO <sub>2</sub> Tool": CO <sub>2</sub> emissions & energy consumption of existing & breakthrough steelmaking routes." Birat, JP., Lorrain, JP., de Lassat, Y.
Birat et al. 1999	"CO <sub>2</sub> Emissions and the Steel Industry's available Responses to the Greenhouse Effect." Birat, JP., Vizioz, JP., de Lassat, Y., Schneider, M., & Jeanneau, M.
Energetics 2005	Steel Industry Marginal Opportunity Study. Energetics, Inc. for U.S. Department of Energy, Industrial Technologies Program. 2005.
Gordon et al. 2010	Ironmaking Technology Selection for Site Specific Conditions. Gordon, Y., Freislich, M., & Els, J.
LBNL 2013	Emerging Energy-efficiency and Carbon Dioxide Emissions-reduction Technologies for the Iron and Steel Industry. Hasanbeigi, A., Price, L., & Arens, M.
Sadoway 2008	"Electrochemical Pathways Towards Carbon-free Metals Production." Sadoway, D.R. Presentation.
UNIDO 2010	Global Technology Roadmap for CCS in Industry: Steel Sectoral Report. Birat, J.P.

Table 6-1. Key Published Sources Reviewed to Identify Practical Minimum Energy Intensities for Select Processes

Numerous fact sheets, case studies, reports, and award notifications were referenced. Details of all of the practical minimum sources consulted can be found in Appendix A4.

Appendix A4 presents details on the R&D technologies that were selected and used to estimate the PM energy intensities. Energy savings from R&D advancements were directly estimated for the six processes. In Appendix A4, technologies are aligned with the most representative process. Some of the technologies have applicability to more than one process (e.g., are crosscutting).

Analysis of the range of energy savings offered by groups of technologies is complicated in that the savings offered by multiple technologies may or may not be additive. Each technology contributes discrete or compounding savings that increase the ultimate savings of the group and some energy savings may be duplicative. As a result, all values are presented as sourced from the literature and energy savings were not aggregated for multiple technologies. A separate study of the individual technologies would be necessary to verify and validate the savings estimates and interrelationships between the technologies. If more than one technology was considered for a particular process, the technology that resulted in the lowest energy intensity was conservatively selected for the PM energy intensity.

R&D in some process areas is more broadly applicable, such as utility/power generation improvements and crosscutting technologies. Cross-cutting technologies applied during the PM analysis included new high-temperature, low-cost ceramic media for natural gas combustion burners, advanced energy and water recovery technology from low-grade waste heat, and control systems for recycling steel residues. The estimated energy savings from crosscutting

improvements were assumed to be applicable to all six processes studied. To calculate PM energy consumption, the CT energy intensity and TM energy intensity were multiplied by the combined estimated savings for crosscutting improvements (1%-16%) and subtracted from the CT energy consumption.

In Appendix A4, the range of technologies considered offer a corresponding range of estimated energy savings. Brief descriptions of the technologies are followed by reported savings in terms of dollars, Btu, and percent savings. The technology developers' estimated savings were taken at face value and adjusted to represent the overall average energy savings potential.

For each technology, Appendix A4 presents a brief explanation of the energy savings and a summary of adjustments necessary to determine the overall average energy savings potential and PM energy intensity. Research savings are speculative in nature. The energy savings will vary depending on the source; they can be reported in terms of primary energy savings, plant-wide energy savings, process energy savings, or energy-type savings. In each case, the reported energy savings were adjusted to determine PM energy intensity.

#### 6.2.1. Weighting of Technologies

The technologies described in Appendix A4 can be weighted differently depending on the audience. Plant managers may primarily be interested in productivity and quality implications; business managers may primarily be interested in relative cost and payback; technology investors may primarily be interested in market impact, technology readiness, and development risk factors; and government regulators may primarily be interested in environmental impacts. Each factor plays heavily into R&D investment considerations.

Appendix A5 (Table A5) considers how to weigh these various perspectives. Six technology weighting factors were considered for each technology:

- A Technology Readiness
- B Market Impact
- C Relative Cost and Savings Payback
- D Technical Risk
- E Productivity/Product Quality Gain
- F Environmental Impacts

Appendix A5 (Table A5) presents the PM technology weighting factors that could be applied to the technologies for specific processes (as identified in Appendix A4). Best engineering judgment was employed to rate each of the technologies with these weighting factors. A score of High, Medium, or Low was assigned to each factor along with a brief explanation for the score. The parameters referenced in scoring are detailed in Appendix A5 (Table A5). An overall importance rating for the technology was determined based on the weighting factor scores. Each weighting factor is assigned a DOE importance level of "1."This importance level can be altered; for example, if Technology Readiness and Market Impact carry higher importance, the

importance level for these factors can be changed to "2" or "3" and the resulting Overall Importance Rating would change accordingly.

The weighting factors presented in Appendix A5 can be used for further study of the R&D technologies identified in Appendix A4. The weighting factor study was part of the analysis of the R&D technologies, and serves as a guide for prioritizing the technologies. However, the weighting factors were not utilized to estimate onsite PM energy intensity or consumption.

# 6.3. PRACTICAL MINIMUM ENERGY CONSUMPTION BY PROCESS AND SECTOR-WIDE

Table 6-2 presents the onsite PM energy consumption for the six processes studied and iron and steel sector-wide. The onsite PM energy savings is the difference between CT energy consumption and PM energy consumption. PM energy savings is equivalent to the sum of *current* and *R&D opportunity* energy savings.

In Table 6-2, data is extrapolated to estimate the total PM subsector opportunity. PM energy consumption for the individual processes studied is summed and the data is extrapolated to estimate sector total. PM subsector energy savings is also expressed as a percent in Table 6-2. This is also shown in Figure 6-2. It is useful to consider both TBtu energy savings and energy savings percent when comparing energy savings opportunity. Both are good measures of opportunity; however, the conclusions are not always the same.

 Table 6-2. Onsite Practical Minimum Energy Consumption, Energy Savings, and Energy Savings Percent for

 the Processes Studied and Sector-Wide

Process	Onsite CT Energy Consumption (TBtu/year)	Onsite PM Energy Consumption (TBtu/year)	PM Energy Savings <sup>†</sup> (CT-PM) (TBtu/year)	PM Energy Savings Percent (CT-PM)/ (CT-TM)*
Integrated Mills				
Agglomeration Sintering	8	6-7	1	17-73%
Cokemaking	36	18-31	4-18	22-91%
Ironmaking Blast Furnace	337	281-329	8-56	9-64%
Steelmaking BOF	20	(-12) - (-10)	30-32	72-77%
Mini Mills				
Steelmaking EAF	101	67-88	13-34	33-86%
Integrated and Mini Mills				
Casting	16	3-4	12-13	74-79%
Rolling Hot Cold	219 86	106-136 32-39	83-113 47-55	38-52% 55-63%
Total for Processes Studied	822	501-625	197-321	39-63%
All Other Processes	176	108-134	42-69	
Total for Iron and Steel Sector- wide	999	609-759	240-390	39-63%

Current typical (CT), Practical minimum (PM), Thermodynamic Minimum (TM)

<sup>†</sup> PM energy savings is the Current Opportunity plus the R&D Opportunity.

\* Calculated using TM from Table 7-2 as the minimum energy of production. This accounts for the energy necessary to perform the process. Potential opportunity reflects the difference between CT and TM energy consumption. Calculation: (CT- PM)/(CT- TM).

\*\* The sector-wide PM energy consumption was an extrapolated value, calculated by dividing the total onsite SOA energy consumption for the processes studied by the overall percent coverage from Table 4-3 (68%).

Figure 6-2 presents the *current opportunity* and the *R&D opportunity* for each process; the *current opportunity* is the difference between CT energy consumption and SOA energy consumption (shown in blue) and the *R&D opportunity* is the difference between the SOA energy consumption and the PM energy consumption (shown in green). In Figure 6-2, the percent savings is the percent of the overall energy consumption bandwidth where TM is the lower baseline. For the processes studied, the greatest *current opportunity* is cokemaking at 69% energy savings. In Figure 6-3, the *current* and *R&D* savings opportunity is shown in terms of TBtu per year savings. The pie chart in Figure 6-3 captures the blue and green portions of the bar chart shown in Figure 6-2, each in a separate pie chart. For the processes studied, the greatest

*current opportunity* in terms of TBtu savings is hot rolling at 83 TBtu per year savings; the greatest *R&D opportunity* in terms of TBtu savings is ironmaking at 249 TBtu per year savings.

To extrapolate the data for all other processes that is shown in Table 6-2 and Figure 6-3, the PM energy consumption of each individual process studied is summed, and the sum is divided by the percent coverage for the entire sector (82%, see Table 4-3). The percent coverage of processes studied compared to the total CT energy consumption of the sector is shown in the last column of Table 4-3. Percent coverage is the ratio of the sum of all the CT energy consumption for the individual processes studied to the CT energy consumption for the sector provided by MECS (see Table 4-3). The PM energy consumption for the remainder of the sector (i.e., all processes that are not included in the six processes studied) was calculated by subtracting the total for the processes studied from the sector-wide total. These additional processes are together referred to as All Other Processes in Table 6-2.

Table 6-2 also presents the PM energy savings percent. To calculate the onsite PM energy savings percent, the thermodynamic minimum (TM) energy consumption serves as the baseline for estimating percent energy savings, not zero. The energy savings percent is the percent of energy saved with PM energy consumption (i.e., the deployment of R&D technologies under development worldwide) compared to CT energy consumption, considering that the TM energy consumption may not be zero (i.e., the TM energy consumption may be negative). As will be explained in Chapter 7, in some cases, the TM reaction energy is a negative value. When comparing energy savings percent from one process to another (or one subsector to another), the absolute savings is the best measure of comparison. The equation for calculating onsite PM energy savings percent is:

 $PM Savings \% = \frac{CT - PM}{CT - TM}$ 



Figure 6-2. Current and R&D Opportunity Energy Savings Bandwidths for the Iron and Steel Processes Studied (with Percent of Overall Energy Consumption Bandwidth)

#### Current Energy Savings Opportunity by Process





\* Extrapolated based on results from processes studied

Figure 6-3. Current and R&D Energy Savings Opportunities by Iron and Steel Process Studied (*Energy Savings per Year in TBtu*)

The PM energy savings opportunity is different than SOA energy savings opportunity in that the scope of the R&D technologies contributing energy savings can essentially be boundless. Putting aside obvious financial, timing, and resource limitations, the process improvements and increased energy efficiency that can be gained through unproven technology is speculative. For this reason, a range is used to represent the potential onsite PM energy consumption, PM energy savings, and PM energy savings percent in Table 6-2. The upper limit of the PM energy consumption range is assumed to be equal to the SOA energy consumption. The lower limit of

the PM energy consumption range was estimated using the method explained in Section 6.2. The lower limit is shown as a dashed line with color fading in the summary figures that present subsector and sector-wide data. This is done because the PM is speculative and depends on unproven R&D technologies; furthermore, the potential energy savings opportunity could be bigger if additional unproven technologies were considered.

# 7. Thermodynamic Minimum Energy Consumption for U.S. Iron and Steel Manufacturing

Real world iron and steel manufacturing does not occur under theoretically ideal conditions; however, understanding the theoretical minimal amount of energy required to manufacture iron and steel can provide a more complete understanding of opportunities for energy savings. This baseline can be used to establish more realistic projections of what R&D energy savings can be achieved. This Chapter presents the thermodynamic minimum (TM) energy consumption required for the processes studied and for the entire sector.

#### 7.1. THERMODYNAMIC MINIMUM ENERGY

TM energy consumption is the calculated minimum amount of energy theoretically needed to complete an iron and steel manufacturing process, assuming ideal conditions that are typically unachievable in real-world applications; in some cases, it is less than zero. TM energy consumption assumes all the energy is used productively and there are no energy losses. It is based on the Gibbs Free Energy ( $\Delta G$ ) equation under ideal conditions for a process. One iron and steel manufacturing process (BOF steelmaking) is theoretically a net producer of energy (i.e., exothermic processes); this created energy was considered in this analysis.

# 7.2. CALCULATED THERMODYNAMIC MINIMUM ENERGY CONSUMPTION FOR INDIVIDUAL PROCESSES

Appendix A1 presents the onsite TM energy consumption for the six processes considered in this bandwidth study in alphabetical order. For a given process, the TM energy intensity is multiplied by the annual U.S. production or throughput to determine the total onsite TM energy consumption (the energy intensity and production/throughput data are also presented in Appendix A1). Table 7-1 presents the references for the TM energy intensity values and the applicable process. Appendix A2 also provides the references for the TM energy intensity data for each individual process. The TM values are mainly derived from earlier work conducted by Carnegie Mellon University for the DOE, and are detailed in the report, Fruehan et al. 2000.

Processes Studied	
Source	Process
de Beer et al. 1998	<i>Future Technologies for Energy-Efficient Iron and Steel Making.</i> de Beer, Worrell, & Blok provide exergy analysis of a steel plant; this report also discusses theoretical values.
Fruehan et al. 2000	Theoretical Minimum Energies to Produce Steel for Selected Conditions. This report highlights minimum values based on theoretical models and specific compositions.
Internal calculations	Calculations based on change in Gibbs free energy at ideal conditions.

 Table 7-1. Published Sources Reviewed to Identify Thermodynamic Minimum Energy Intensities for the

 Processes Studied

BOF steelmaking can at times result in net energy gain through exothermic processes. For exothermic iron and steel manufacturing processes, a zero baseline would result in negative percent savings, a physical impossibility. TM energy consumption was instead referenced as the baseline (or minimum amount of energy) when calculating the absolute energy savings potential. The equations used to determine the absolute energy savings for SOA and PM are as follows:

SOA Savings 
$$\% = \frac{CT - SOA}{CT - TM}$$
  
PM Savings  $\% = \frac{CT - PM}{CT - TM}$ 

For processes requiring an energy intensive transformation (e.g., ironmaking or EAF steelmaking), this percent energy savings approach results more realistic and comparable energy savings estimates. Using zero as the baseline (or minimum amount of energy) would exaggerate the total bandwidth to which SOA energy savings and PM energy savings are compared to determine the energy savings percent. When TM energy consumption is referenced as the baseline, SOA energy savings and PM energy savings are relatively more comparable, resulting in more accurate energy savings percentages.

TM energy intensity is the least amount of energy required for each of the six process areas examined in this report – ore agglomeration, cokemaking, ironmaking, steelmaking, casting, and rolling. The full credit of off-gas chemical and thermal energy is considered. The discussion below provides a brief summary of TM results by process area from Fruehan et al. 2000. Some of the tables from Fruehan et al. 2000 are provided in Appendix A3, including some additional references not specifically sourced in this report. For further details, readers are encouraged to review the full source.

#### 7.2.1. Ore Agglomeration and Cokemaking

Ore agglomeration consists of taking ore fines or concentrate and sintering them into sinter or pellets. Cokemaking is achieved by heating coal under pressure, driving off volatiles and increasing the strength of the material. The theoretical minimum energies for these processes were determined by calculating the energy to heat idealized coal and ore to 1100°C and 1350°C, respectively. In cokemaking the heat of devolatilization is also required but is usually accounted for in computing the energy value of coal. The coke output was taken as 0.768 per metric ton of the coal input. The volatiles represent the remainder, which is used as a fuel or for chemicals (Fruehan et al. 2000). The TM for sintering is approximately 1.03 MMBtu/ton (1.20 GJ/tonne) of sinter; and the TM for cokemaking is approximately 1.72 MMBtu/ton (2.00 GJ/tonne) of coke output.

#### 7.2.2. Ironmaking

For iron ore reduction, Fruehan et al. 2000's base calculations considered pure  $Fe_2O_3$  reduced at 298 Kelvin (K), heated to melting point and melted (no superheat); resulting in a TM value of

7.4 MMBtu/ton (8.6 GJ/tonne) of hot metal. In blast furnace ironmaking, the calculated TM base case is 8.43 MMBtu/ton (9.81 GJ/tonne) of hot metal – with iron nearly saturated with carbon. Other cases include effects of impurities, slags, and coke ash – with TM values ranging from 8.8-9.0 MMBtu/ton (10.2-10.4 GJ/tonne). In all cases, full credit is assumed for the energy in the offgas.

For direct reduced iron where reduction using natural gas occurs without melting, pure  $Fe_2O_3$  can be produced with 7.19 MMBtu/ton (8.37 GJ/tonne). Other cases presented by Fruehan et al. 2000 result in TM values ranging from 6.8-8.1 MMBtu/ton (7.9-9.3 GJ/tonne).

## 7.2.3. Steelmaking

Traditional integrated oxygen steelmaking produces energy as carbon and other elements are oxidized. Inputs of one ton of hot metal plus 0.29 tons of cold scrap yield 1.24 tons liquid steel with a TM energy requirement of 6.8 MMBtu/ton (7.9 GJ/tonne) of liquid steel (including ironmaking energy). Fruehan et al. 2000 notes potential energy in off gas, with a net value of 0.6 MMBtu/ton (0.8 GJ/tonne) of steel.

For producing steel from pure scrap, the metal is heated and melted (without superheat) resulting in a calculated TM value of 1.1 MMBtu/ton (1.3 GJ/tonne). Other cases presented by Fruehan et al. 2000 result in TM values ranging from 1.1-1.4 MMBtu/ton (1.3-1.6 GJ/tonne).

Fruehan et al. 2000 also provides TM values for the production of stainless steel with high chromium and nickel content. For a typical input mix in an EAF, the TM value for stainless steel of approximately 1.0 MMBtu/ton (1.2 GJ/tonne) is about 9% less than the pure iron TM value.

## 7.2.4. Casting and Rolling

Various estimates for hot rolling and cold rolling were provided and consider both heated and unheated slabs, normal slabs and thin slabs, as well as bars. Fruehan et al. 2000 presents values for both heat energy and deformation energy (electricity). For unheated slabs, the TM heating value for hot rolling is approximately 0.71 MMBtu/ton (0.83 GJ/tonne); and approximately 0.23 MMBtu/ton (0.27 GJ/tonne) when heating from 1173 K to 1473 K; and the deformation energy ranges from 0.01 MMBtu/ton to 0.06 MMBtu/ton (0.01 to 0.07 GJ/tonne) for rolling depending on the product and conditions. For cold rolling Fruehan et al. 2000 presents values for cold rolling process only, and does not include any additional operations (as is the case with CT, SOA, and PM cold rolling consumption estimates).

# 7.3. THERMODYNAMIC MINIMUM ENERGY CONSUMPTION BY PROCESS AND SECTOR-WIDE

The minimum baseline of energy consumption for an iron and steel manufacturing process is its TM energy consumption. If all the 2010 level of iron and steel production occurred at TM energy intensity, there would be 100% savings. The percentage of energy savings is determined by

calculating the absolute decrease in energy consumption and dividing it by the total possible savings (CT energy consumption-TM energy consumption).

Table 7-2 provides the TM energy consumption for the six processes studied (excluding feedstock energy). In theory, if heat generating processes could be carefully coupled with heat consuming processes, this could greatly offset the energy usage in iron and steel manufacturing overall. It is an imperative to keep in mind that ideal conditions are largely unrealistic goals in practice and these values serve only as a guide to estimating energy savings opportunities.

Table 7-2 also presents the extrapolated TM energy consumption for the entire sector. The extrapolation for sector-wide TM energy consumption is done with the same methodology as for SOA energy consumption and PM energy consumption (as explained in Section 5.2 and 6.4).

The TM energy consumption was used to calculate the *current* and *R&D* energy savings percentages (not zero).

Table 7-2. Thermodynamic Minimum Energy Consumption by Process and           Sector-Wide for the Six Processes Studied and Extrapolated to Sector Total		
Process	Onsite TM Energy Consumption (TBtu/year)	
Integrated Mills		
Agglomeration Sintering	6	
Cokemaking	16	
Ironmaking Blast Furnace	250	
Steelmaking BOF	-22	
Mini Mills		
Steelmaking EAF (Pure Scrap)	62	
Integrated and Mini Mills		
Casting	<0.01	
Rolling Hot Cold	2 <1	
Total for Processes Studied	313	
All Other Processes	67	
Total for Iron and Steel Sector-wide     381		

Thermodynamic minimum (TM)

*†* Estimates for the entire sector were extrapolated by dividing the onsite TM energy consumption for the processes studied by the overall percent coverage of 82% (see Table 4-3).

# 8. U.S. Iron and Steel Manufacturing Energy Bandwidth Summary

This Chapter presents the energy savings bandwidths for the iron and steel manufacturing processes studied and sector-wide based on the analysis and data presented in the previous Chapters and the Appendices. Data is presented for the six processes studied and extrapolated to estimate the energy savings potential for all of U.S. iron and steel.

# 8.1. IRON AND STEEL BANDWIDTH PROFILE

Table 8-1 presents the *current opportunity* and *R&D opportunity* energy savings for the six processes studied and extrapolated to estimate the sector total. The process totals are summed to provide a sector-wide estimate. The energy savings data was extrapolated to account for all other processes not included in the nine processes studied, as explained in Section 5.2 (SOA) and 6.4 (PM). Each row in Table 8-1 shows the opportunity bandwidth for a specific iron and steel manufacturing process and sector-wide.

As shown in Figure 8-1, four hypothetical opportunity bandwidths for energy savings are estimated (as defined in Chapter 1). To complete the six processes studied, the analysis shows the following:

- *Current Opportunity* 197 TBtu per year of energy savings could be obtained if state of the art technologies and practices are deployed.
- *R&D Opportunity* 124 TBtu per year of additional energy savings could be attained in the future if applied R&D technologies under development worldwide are deployed (i.e., reaching the practical minimum).

To complete all of the U.S. iron and steel sector processes (based on extrapolated data), the analysis shows the following:

- *Current Opportunity* 240 TBtu per year of energy savings could be obtained if state of the art technologies and practices are deployed.
- *R&D Opportunity* 150 TBtu per year of additional energy savings could be attained in the future if applied R&D technologies under development worldwide are deployed (i.e., reaching the practical minimum).

Figure 8-1 also shows the estimated *current* and *R&D* energy savings opportunities for individual iron and steel manufacturing processes. The U.S. iron and steel industry operated at relatively low capacity utilization and lower-than-typical efficiencies in 2010, due in large part to the economic downturn. While the specific impacts of the economic factors in 2010 are not directly identified in this report, it is reasonable to assume that the current opportunity is likely somewhat exaggerated, as a portion of the current savings could be achieved by simply optimizing production rates. For this reason the border between *current opportunity* and *R&D opportunity* is

not explicitly defined, and a dashed line and color fading is used. The area between R&Dopportunity and impractical is shown as a dashed line with color fading because the PM energy savings impacts are speculative and based on unproven technologies.

Process	Current Opportunity (CT-SOA) (TBtu/year)	R&D Opportunity (SOA-PM) (TBtu/year)
ntegrated Mills		
Agglomeration Sintering	<1	1
Cokemaking	4	13
Ironmaking Blast Furnace	7	49
Steelmaking BOF	30	2
Mini Mills		
Steelmaking EAF (Pure Scrap)	13	21
ntegrated and Mini Mills		
Casting	12	1
Rolling Hot Cold	83 47	30 7
Total for Processes Studied	197	124
All Other Processes	42	27
Total for Iron and Steel Sector-wide	240	150

T 1 1- 0 4 0 

Current typical (CT), state of the art (SOA), practical minimum (PM)

From the processes studied, the greatest *current* energy savings opportunity for iron and steel manufacturing comes from upgrading production methods in hot rolling. The greatest R&D energy savings opportunity for iron and steel comes from blast furnace ironmaking.

The *impractical* bandwidth represents the energy savings potential that would require fundamental changes in iron and steel manufacturing. It is the difference between PM energy consumption and TM energy consumption. The term *impractical* is used because the significant research investment required based on today's knowledge would no longer be practical because of the thermodynamic limitations. The TM energy consumption is based on ideal conditions that are typically unattainable in commercial applications. It was used as the baseline for calculating the energy savings potentials (not zero) to provide more accurate targets of energy savings opportunities.




Figure 8-2 shows the bandwidth summaries for the iron and steel processes presented in order of highest current plus R&D energy savings opportunity. Blast furnace ironmaking is the largest energy consuming process in iron and steel manufacturing. If the lower limit of PM energy consumption could be reached, this would save about 56 TBtu/year compared to CT, amounting to 6% of CT energy consumption for the entire iron and steel sector. Hot rolling, the second largest energy consuming process in iron and steel manufacturing, would save about 113 TBtu copared to CT if the lower limit of PM energy consumption. Other processes, such as cokemaking, casting, and agglomeration, have a much smaller difference between CT energy consumption and the PM energy consumption.



Figure 8-2. Current and R&D Opportunity Energy Savings Bandwidths for the Processes Studied (with Percent of Overall Energy Consumption Bandwidth)

# 9. References

AISI 2011a	2010 Annual Statistical Report. American Iron and Steel Institute (AISI). 2011.
AISI 2011b	Profile of the American Iron and Steel Institute 2010-2011. American Iron and Steel Institute (AISI). 2011. www.steel.org/~/media/Files/AISI/About%20AISI/Profile%20Brochure%20F- singles_CX.pdf
AISI 2012	Profile of the American Iron and Steel Institute 2012. American Iron and Steel Institute (AISI). 2012. www.steel.org/~/media/Files/AISI/Reports/2012%20Profile%20F4-SDMIcrx- sm.pdf
AISI 2013a	"2013 Steel Industry Profile." American Iron and Steel Institute (AISI). 2013. www.steel.org/About%20AISI/Industry%20Profile.aspx
AISI 2013b	"Steelmaking Flowlines." American Iron and Steel Institute (AISI). 2013. www.steel.org/Making%20Steel/How%20Its%20Made/Steelmaking%20Flowlin es.aspx
AIST 2011	2011 Electric Arc Furnace Roundup. Association for Iron & Steel Technology (AIST). 2011.
ArcelorMittal 2014	"2013 ArcelorMittal USA Fact Book." ArcelorMittal. 2014. www.usa.arcelormittal.com/Documents/ArcelorMittal-USA/Publications- reports/2013FactBook.pdf
Birat et al. 1999	"CO <sub>2</sub> Emissions and the Steel Industry's available Responses to the Greenhouse Effect." Birat, JP., Vizioz, JP., de Lassat, Y., Schneider, M., & Jeanneau, M. <i>IRSID</i> . 1999. <u>www.ulcos.org/en/docs/Ref16%20-%20Sdiego.pdf</u>
Birat et al. 2009	"The "CO <sub>2</sub> Tool": CO <sub>2</sub> emissions & energy consumption of existing & breakthrough steelmaking routes." Birat, JP., Lorrain, JP., de Lassat, Y. <i>La Revue de Metallurgie</i> , <i>106</i> (09): 325-336.
Considine 2012	<i>Economic Impacts of the American Steel Industry</i> . Considine, T.J. 2012. <u>http://legacy.steel.org/news/NewSteelNews/images/PDFs/Considine_March%20</u> <u>2012.pdf</u>
de Beer et al. 1998	<i>Future Technologies for Energy-Efficient Iron and Steel Making.</i> de Beer, J., Worrell, E., & Blok, K. <i>Annu. Rev. Energy Environ.</i> 23:123-205. 1998.

DOE 2013	"Energy Savings Assessments (ESA)." Conducted by the U.S. Department of Energy (DOE) and Oak Ridge National Laboratory (ORNL) between 2006 and 2011. Results obtained from personal communications with Sachin Nimbalkar, Oak Ridge National Laboratory.
DOE 2014	<i>Manufacturing Energy and Carbon Footprints</i> . U.S. Department of Energy (DOE), Advanced Manufacturing Office (AMO). 2014. www.energy.gov/eere/amo/manufacturing-energy-and-carbon-footprints-2010- mecs
EIA 2009	"Manufacturing Energy and Consumption Survey (MECS) 2006." U.S. Energy Information Administration (EIA). 2009. www.eia.gov/consumption/manufacturing/data/2006/
EIA 2013a	"Manufacturing Energy and Consumption Survey (MECS) 2010." U.S. Energy Information Administration (EIA). 2013. www.eia.gov/consumption/manufacturing/data/2010/
EIA 2013b	Model Documentation Report: Industrial Demand Module of the National Energy Modeling System. U.S. Energy Information Administration (EIA). 2013. www.eia.gov/forecasts/aeo/nems/documentation/industrial/pdf/m064%282013% 29.pdf
Energetics 2000	<i>Energy and Environmental Profile of the U.S. Steel Industry</i> . Energetics, Inc. for U.S. Department of Energy (DOE), Office of Industrial Technologies (OIT). 2000. <u>www.energy.gov/sites/prod/files/2013/11/f4/steel_profile.pdf</u>
Energetics 2004	<i>Steel Industry Energy Bandwidth Study</i> . Energetics, Inc. for U.S. Department of Energy (DOE), Industrial Technologies Program (ITP). 2004. <u>www.energy.gov/sites/prod/files/2013/11/f4/steelbandwidthstudy.pdf</u>
Energetics 2005	<i>Steel Industry Marginal Opportunity Study</i> . Energetics, Inc. for U.S. Department of Energy (DOE), Industrial Technologies Program (ITP). 2005. <a href="https://www.energy.gov/sites/prod/files/2013/11/f4/steelmarginalopportunity.pdf">www.energy.gov/sites/prod/files/2013/11/f4/steelmarginalopportunity.pdf</a>
Energiron 2013	"Energiron: The Innovative Direct Reduction Technology." Energiron. 2013. www.energiron.com/
EPA 2011	"Location of U.S. Facilities." U.S. Environmental Protection Agency (EPA). 2011. <u>www.epa.gov/sectors/sectorinfo/sectorprofiles/ironsteel/map.html</u>
EPA 2012	Available and Emerging Technologies for Reducing Greenhouse Gas Emissions from the Iron and Steel Industry. U.S. Environmental Protection Agency (EPA). 2012. <u>www.epa.gov/nsr/ghgdocs/ironsteel.pdf</u>

Fruehan et al. 2000	<i>Theoretical Minimum Energies to Produce Steel for Selected Conditions.</i> Fruehan, R.J., Fortini, O., Paxton, H.W., and Brindle, R. for the U.S. Department of Energy (DOE), Office of Industrial Technologies (OIT). 2000. www.energy.gov/sites/prod/files/2013/11/f4/theoretical_minimum_energies.pdf
Giavani et al. 2012	"Consteel® Evolution <sup>TM</sup> – The Second Generation of Consteel Technology." Giavani, C., Malfa, E., & Memoli, F. <i>Industrial Heating</i> . 2012. www.industrialheating.com/articles/90594-consteel-evolutionthe-second- generation-of-consteel-technology
Gordon et al. 2010	Ironmaking Technology Selection for Site Specific Conditions. Gordon, Y., Freislich, M., & Els, J. AISTech 2010 Proceedings – Volume I: 519-528. 2010.
Griggs 2013	"Nucor's La. plant near completion." Griggs, T. <i>The Advocate</i> . 2013. www.theadvocate.com/home/6552645-125/nucors-la-plant-near-completion
IEA 2007	Tracking Industrial Energy Efficiency and CO <sub>2</sub> Emissions. International Energy Agency (IEA). 2007. www.iea.org/publications/freepublications/publication/tracking_emissions.pdf
IPPC 2013	Best Available Techniques (BAT) Reference Document for Iron and Steel Production. European Commission. Integrated Pollution Prevention and Control (IPPC). 2013. http://eippcb.jrc.ec.europa.eu/reference/BREF/IS_Adopted_03_2012.pdf
LBNL 2000	<i>Emerging Energy-Efficient Industrial Technologies</i> . Martin, N., Worrell, E., Ruth, M., Price, L., Elliott, R.N., Shipley, A.M., & Thorne, J. for Lawrence Berkeley National Laboratory (LBNL). 2000. www.osti.gov/scitech/servlets/purl/920147-u4MkO8/
LBNL 2008	World Best Practice Energy Intensity Values for Selected Industrial Sectors. Worrell, E., Price, L., Neelis, M., Galitsky, C., & Nan, Z. for Lawrence Berkeley National Laboratory (LBNL). 2008. http://eetd.lbl.gov/sites/all/files/industrial_best_practice_en.pdf
LBNL 2010a	Development of Bottom-up Representation of Industrial Energy Efficient Technologies in Integrated Assessment Models for the Iron and Steel Sector. Xu, T., Sathaye, J., & Galitsky, C. for Lawrence Berkeley National Laboratory (LBNL). 2010. <u>http://ies.lbl.gov/sites/all/files/lbnl-4314e_0.pdf</u>
LBNL 2010b	Energy Efficiency Improvement and Cost Saving Opportunities for the Iron and Steel Industry: An ENERGY STAR Guide for Energy and Plant Managers. Neelis, M., Worrell, E., & Masanet, E. for Lawrence Berkeley National Laboratory (LBNL). 2010. <u>http://ies.lbl.gov/sites/all/files/lbnl-4779e_0.pdf</u>

LBNL 2013	Emerging Energy-efficiency and Carbon Dioxide Emissions-reduction Technologies for the Iron and Steel Industry. Hasanbeigi, A., Price, L., & Arens, M. for Lawrence Berkeley National Laboratory (LBNL). 2013. <u>http://china.lbl.gov/sites/all/files/6106e-steel-tech.pdf</u>				
McGannon 1971	The Making, Shaping and Treating of Steel, 9 <sup>th</sup> ed. McGannon, H. E. 1971.				
NRC 2007	Benchmarking Energy Intensity in the Canadian Steel Industry. Prepared by the Canadian Industry Program for Energy Conservation for Canadian Steel Producers Association and Natural Resources Canada (NRC). 2007. www.nrcan.gc.ca/sites/oee.nrcan.gc.ca/files/pdf/industrial/technical- info/benchmarking/csi/documents/SteelBenchmarkEnglish.pdf				
NSF 2007	"Table 50: Funds for industrial R&D as a percent of net sales of companies performing industrial R&D in the United States, by industry and company size: 2007." <i>Industrial Research and Development Information System</i> . National Science Foundation (NSF). www.nsf.gov/statistics/iris/tables.cfm?pub_year=nsf_2007				
Sadoway 2008	"Electrochemical Pathways Towards Carbon-free Metals Production." Sadoway, D.R. Presentation. 2008. <u>http://gcep.stanford.edu/pdfs/2RK4ZjKBF2f71uM4uriP9g/SadowayGCEP_reduced.pdf</u>				
SRI 2013a	"Steel Recycling Institute." Steel Recycling Institute (SRI). 2013. <u>www.recycle-steel.org</u>				
SRI 2013b	"Steel Recycling Rates." Steel Recycling Institute (SRI). 2013. <u>www.recycle-steel.org/Recycling%20Resources/Steel%20Recycling%20Rates.aspx</u>				
Steel Dynamics, Inc. n.d.	Welcome to Mesabi Nugget. Presentation. n.d. www.mesabinuggetmn.com/pdf/mesabinuggetoverview.ppt				
Stubbles 2000	Energy Use in the U.S. Steel Industry: An Historical Perspective and Future Opportunities. Stubbles, J. for the U.S. Department of Energy, Office of Industrial Technologies. 2000. www.energy.gov/sites/prod/files/2013/11/f4/steel_energy_use.pdf				
UNIDO 2010	<i>Global Technology Roadmap for CCS in Industry: Steel Sectoral Report</i> . Birat, JP for the United Nations Industrial Development Organization (UNIDO). 2010. http://decarboni.se/sites/default/files/publications/15671/global-technology-roadmap-ccs-industry-steel-sectoral-report.pdf				
USCB 2012	"North American Industry Classification System." United States Census Bureau (USCB). 2012. <u>www.census.gov/eos/www/naics/index.html</u>				

USGS 2012a	2010 Minerals Yearbook: Iron and Steel [Advance Release]. U.S. Geological Survey (USGS). 2012. http://minerals.usgs.gov/minerals/pubs/commodity/iron_&_steel/myb1-2010- feste.pdf
USGS 2012b	<i>Mineral Commodity Summaries, January 2012: Iron and Steel.</i> Prepared by Fenton, M.D. for the U.S. Geological Survey (USGS). 2012. <u>http://minerals.er.usgs.gov/minerals/pubs/commodity/iron_&amp;_steel/mcs-2012-feste.pdf</u>
USGS 2012c	2010 Minerals Yearbook: Iron and Steel Scrap [Advance Release]. U.S. Geological Survey (USGS). 2012. http://minerals.usgs.gov/minerals/pubs/commodity/iron_%26_steel_scrap/myb1- 2010-fescr.pdf
Voestalpine 2013	"Voestalpine constructing direct reduction plan in Texas, USA." Voestalpine. 2013. <u>www.voestalpine.com/group/en/press/press-releases/2013-03-13-</u> voestalpine-constructing-direct-reduction-plant-in-texas-usa.html
WCA 2013	"Coal & Steel." World Coal Association (WCA). 2013. www.worldcoal.org/coal/uses-of-coal/coal-steel
WSA 2012	Sustainable Steel: At the core of a green economy. World Steel Association (WSA). 2012. www.worldsteel.org/dms/internetDocumentList/bookshop/Sustainable-steel-at- the-core-of-a-green-economy/document/Sustainable-steel-at-the-core-of-a-green- economy.pdf

### **Appendix A1: Master Iron and Steel Table**

 Table A1. U.S. Production Volume of Six Iron and Steel Processes in 2010 with Energy Intensity Estimates and Calculated Onsite Energy

 Consumption for the Four Bandwidth Measures (Excludes Feedstock Energy)

	2010	Energy Intensity (MMBtu/ton) [GJ/tonne]			Calculated Onsite Energy Consumption (TBtu/year)				
Process	Production (1,000 tons)	CY	SOA	PM Lower Limit	тм"	СҮ	SOA	PM Lower Limit	ТМ
Agglomeration Pelletizing <sup>a</sup> Sintering	39,728 5,759	0.70 [0.82] 1.32 [1.54]	- 1.27 [1.48]	- 1.11 [1.29]	- 1.03 [1.20]	7.60	7.31	6.39	5.94
Cokemaking	9,292	3.83 [4.46]	3.37 [3.92]	1.92 [2.23]	1.72 [2.00]	35.59	31.31	17.84	15.98
Ironmaking Blast Furnace Direct Reduction <sup>b</sup>	29,590 0	11.72 [13.63] 9.17 [10.71]	11.13 [12.95] 8.33 [9.68]	9.49 [11.04] _ <sup>d</sup>	8.43 [9.81] 7.19 [8.36]	336.79 <sup>e</sup> 0	329.34 0	280.81 0	249.47 0
Steelmaking BOF-based EAF	34,345 54,386	0.58 [0.67] 1.86 [2.16]	-0.30 [-0.35] 1.62 [1.88]	-0.36 [-0.42] 1.24 [1.44]	-0.64 [-0.74] 1.14 [1.33]	19.92 101.16	-10.30 88.11	-12.36 67.44	-22.15 62.06
Casting	84,784	0.19 [0.22]	0.05 [0.06]	0.04 [0.05]	<0.01	16.11	4.24	3.39	<0.01
Rolling Hot <sup>°</sup>	84,784	2.58 [3.00]	1.60 [1.86]	1.25 [1.45]	0.02-0.25 [0.02-0.29]	218.74	135.65	105.98	1.70
Cold	27,710	3.48 [4.05]	1.42 [1.65]	1.15 [1.34]	0.02 [0.02]	86.43 <sup>e</sup>	39.35	31.87	0.41

<sup>a</sup> Conducted primarily at mining site, which is outside the scope of the analysis and thus only shown for reference

<sup>b</sup> Traditional direct reduction processes not utilized in the U.S. in 2010; a small amount of iron nuggets (approximately 75 thousand tons) were produced by the new ITmk3 process in start-up operations at the first commercial plant in Minnesota, associated energy data not available

<sup>c</sup> TM range for hot rolling depends on whether some heating is assumed or not

<sup>d</sup> No value currently available

<sup>e</sup> Current typical values for blast furnace and cold rolling each exclude 10 TBtu of nonfuel feedstock natural gas.

The four bandwidth measures are current typical (CT), state of the art (SOA), practical minimum (PM), and thermodynamic minimum (TM).

# Appendix A2: References for U.S. Production Data of the 6 Processes Studied and Energy Intensity Data Used to Calculate the Current Typical, State of the Art, and Thermodynamic Minimum Energy Consumption Bands

Table A2. References for U.S. Throughput Data of the Six Iron and Steel Processes Studied and Energy Intensity Data Used to Calculate the Current Typical. State of the Art, and Thermodynamic Minimum Energy Consumption Bands **CT Energy Intensity** SOA Energy Intensity Production Process **TM Energy Intensity Reference** Reference Reference(s) Reference(s) Agglomeration Pelletizing: IPPC 2013 Pelletizing AISI 2011a Sintering: LBNL 2008 Fruehan et al. 2000 Sintering: IPPC 2013 Sintering AISI 2011a NRC 2007 Fruehan et al. 2000 Cokemaking AISI 2011a: EIA 2013b Ironmaking Blast Furnace: NRC 2007 Blast Furnace AISI 2011a AISI 2011a; IPPC 2013 Direct Reduction: Energiron Fruehan et al. 2000 Direct Reduction 2013 BOF-based: NRC 2007 Steelmaking BOF-based: IPPC 2013 **BOF-based** AISI 2011a EAF: NRC 2007: Giavani et Fruehan et al. 2000 EAF: AIST 2011 EAF al. 2012 Casting AISI 2011a NRC 2007 NRC 2007; LBNL 2008 Fruehan et al. 2000 Rolling Hot: NRC 2007 Hot: NRC 2007 Hot AISI 2011a Fruehan et al. 2000 Cold: EIA 2013b Cold: LBNL 2008 Cold

The four bandwidth measures are current typical (CT), state of the art (SOA), practical minimum (PM), and thermodynamic minimum (TM).

# Appendix A3: CT, SOA, and TM Energy Intensities by Process and Energy Type

### **CURRENT TYPICAL ENERGY INTENSITIES**

Table A3-1. Current Typical Energy Intensity Estimate for Pelletizing					
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference		
Natural Gas	0.01	0.01	IPPC 2013 (Table 4.1 and text, converted to HHV, see A3-27)		
COG/BFG	0.30	0.35	IPPC 2013 (Table 4.1 and text, converted to HHV)		
Coke and Breeze	0.31	0.36	IPPC 2013 (Table 4.1 and text, converted to HHV)		
Electricity	0.09	0.10	IPPC 2013 (Table 4.1 assumes worst case)		
Total Energy/ Unit	0.70	0.82			

#### Table A3-2. Current Typical Energy Intensity Estimate for Sintering

Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Natural Gas	0.07	0.08	IPPC 2013 (Table 3.2, converted to HHV)
Coke and Breeze	1.17	1.36	IPPC 2013 (Table 3.2, converted to HHV)
Electricity	0.09	0.10	IPPC 2013 (Table 3.2, converted to onsite energy)
Total Energy/ Unit	1.32	1.54	

Table A3-3. Current Typical Energy Intensity Estimate for Cokemaking						
Fuel Source	Energy Intensity (including merchant coke) (MMBtu/ton)*	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference		
Natural Gas	0.26	0.33	0.39	Calculated based on production and AISI 2011a Table 37		
COG	2.15	2.78	3.24	Calculated based on production and AISI 2011a Table 37		
BFG	0.03	0.03	0.04	Calculated based on production and AISI 2011a Table 37		
Steam	0.44	0.57	0.66	Estimate based on EIA 2013b		
Electricity	0.08	0.10	0.12	Estimate based on EIA 2013b		
Total Energy/ Unit	2.96	3.83	4.45			

\* Energy intensity per ton of total coke consumed (including offsite purchased merchant coke) is <u>not</u> referenced in determining CY. Natural gas, COG, BFG energy intensity is converted to ton of coke produced using the ratio of 9,292,000 tons coke produced to 12,021,000 coke consumed (AISI 2011a, Table 31).

Table A3-4. Current Typical Energy Intensity Estimate for Blast Furnace Ironmaking						
Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference			
Coke	10.54	12.26	Calculated based on production and AISI 2011a Table 31			
Natural Gas	1.62	1.88	Calculated based on production and AISI 2011a Table 37			
Oil	0.14	0.16	Calculated based on production and AISI 2011a Table 37			
Coal	1.83	2.13	Calculated based on production and AISI 2011a Table 29			
Steam	0.04	0.05	Based on IPPC 2013			
Electricity	0.23	0.27	Based on IPPC 2013 (converted to onsite energy)			
COG	0.24	0.28	Calculated based on production and AISI 2011a Table 37			
BFG (Net)	-2.93	-3.40	Calculated based on production and AISI 2011a Table 37			
Total Energy/ Unit	11.72	13.63				

### Table A3-4. Current Typical Energy Intensity Estimate for Blast Furnace Ironmaking

Table A3-5. Current Typical Energy Intensity Estimate for Basic Oxygen Furnace Steelmaking						
Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference			
Natural Gas/ COG	0.37	0.43	Based on IPPC 2013 (converted to HHV)			
BFG	0.03	0.03	Based on IPPC 2013 (converted to HHV)			
Steam	0.07	0.08	Based on IPPC 2013			
Electricity	0.11	0.13	Based on IPPC 2013 (converted to onsite energy)			
Total Energy/ Unit	0.58	0.67				

#### Table A3-6. Current Typical Energy Intensity Estimate for Electric Arc Furnace Steelmaking

Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Natural Gas	0.44	0.51	Estimated from AIST 2011
Electricity	1.42	1.65	Estimated from AIST 2011
Total Energy/ Unit	1.86	2.16	

Table A3-7. Current Typical Energy Intensity Estimate for Direct Reduced Ironmaking					
Fuel Source         Energy Intensity (MMBtu/ton)         Energy Intensity (GJ/MT)         Reference					
Natural Gas	8.9	10.4	IEA 2007, Chapter 5		
Electricity	0.27	0.31	IEA 2007, Chapter 5		
Total Energy/ Unit	9.17	10.71			

#### Table A3-8. Current Typical Energy Intensity Estimate for Continuous Casting

Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Natural Gas	0.08	0.09	Based on NRC 2007 Figure 4-12 (converted to HHV)
Electricity	0.07	0.08	Based on NRC 2007 Figure 4-12 (converted to onsite energy)
Plant Utilities	0.04	0.05	Based on NRC 2007 Figure 4-12 (converted to onsite energy)
Total Energy/ Unit	0.19	0.22	

Table A2-0 Current Ty	mical Enorgy Intonsit	y Estimate for Hot Rolling	
I able A3-3. Guilent I	pical Ellergy intensit		

······································						
Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference			
	Hot Strip and Plate Mills					
Natural Gas	2.55	2.96	Based on NRC 2007 Figure 4-14 (converted to HHV)			
Plant Utilities	0.08	0.09	Based on NRC 2007 Figure 4-14 (electrical, converted to onsite energy)			
Electricity	0.40	0.47	Based on NRC 2007 Figure 4-14 (converted to onsite energy)			
Recovery (Credit)	-0.03	-0.04	Based on NRC 2007 Figure 4-14			
Total Energy/ Unit	2.99	3.48				
		Section Mills				
Natural Gas	1.79	2.08	Based on NRC 2007 Figure 4-16 (converted to HHV)			
Plant Utilities	0.03	0.03	Based on NRC 2007 Figure 4-16 (assumed to be electrical, converted to onsite energy)			
Electricity	0.34	0.40	Based on NRC 2007 Figure 4-16 (converted to onsite energy)			
Total Energy/ Unit	2.16	2.51				
	Average for H	ot Strip, Plate and Se	ection Mills *			
Natural Gas	2.17	2.52				
Plant Utilities, Recovery	0.03	0.04				
Electricity	0.38	0.44				
Total Energy/ Unit	2.58	3.00				

\* Average referenced for Hot Rolling CA

Table A3-10. Current Typical Energy Intensity Estimate for Cold Rolling				
Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference	
Natural Gas	1.58	1.84	EIA 2013b Table B8	
Distillate	0.01	0.01	EIA 2013b Table B8	
Steam	0.85	0.99	EIA 2013b Table B8	
Electricity	1.05	1.22	EIA 2013b Table B8	
Total Energy/ Unit	3.48	4.05		

### **STATE OF THE ART ENERGY INTENSITIES**

Table A3-11. State of the Art Energy Intensity Estimate for Sintering				
Fuel Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference	
Natural Gas	0.06	0.07	Based on LBNL 2008, Table 2.1.3 and text (converted to HHV, see A3-28)	
Coke and Breeze	1.12	1.30	Based on LBNL 2008, Table 2.1.3 and text (converted to HHV)	
Electricity	0.09	0.10	Based on LBNL 2008, Table 2.1.3 and text (converted to onsite energy)	
Total Energy/ Unit	1.27	1.47		

#### Table A3-12. State of the Art Energy Intensity Estimate for Cokemaking

Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Natural Gas	0.33	0.38	Based on NRC 2007 (converted to HHV)
COG	2.72	3.16	Based on NRC 2007 (converted to HHV)
BFG	0.03	0.04	Based on NRC 2007 (converted to HHV)
Steam	0.22	0.25	Based on NRC 2007
Electricity	0.09	0.10	Based on NRC 2007 (converted to onsite energy)
Total Energy/ Unit	3.37	3.92	

Table A3-13. State of the Art Energy Intensity Estimate for Blast Furnace Ironmaking			
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Coke	9.88	11.49	Based on NRC 2007 (converted to HHV)
Oil	1.21	1.41	Based on NRC 2007 (converted to HHV)
Coal	2.31	2.69	Based on NRC 2007 (converted to HHV)
Steam	0.34	0.40	Based on NRC 2007
Electricity	0.08	0.09	Based on NRC 2007 (converted to onsite energy)
COG	0.22	0.26	Based on NRC 2007 (converted to HHV)
BFG (Net)	-2.92	-3.40	Based on NRC 2007 (converted to HHV)
Total Energy/ Unit	11.13	12.94	

**Energy Intensity Energy Intensity Energy Source** Reference (MMBtu/ton) (GJ/MT) Based on NRC 2007 (converted to onsite Electricity 0.08 0.09 energy) Based on NRC 2007 (assumed to be natural Other 0.34 0.40 gas, converted to HHV) Steam (Credit) -0.15 -0.17 Based on NRC 2007 Based on NRC 2007 (converted to HHV) BOF Gas (Credit) -0.67 -0.58 **Total Energy/ Unit** -0.30 -0.35

Table A3-15. State of the Art Energy Intensity Estimates for Electric Arc Furnace Steelmaking				
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference	
Electricity	1.47	1.71	Based on NRC 2007 (converted to onsite energy)	
Natural gas	0.15	0.18	Based on NRC 2007 (converted to HHV)	

			energy,
Natural gas	0.15	0.18	Based on NRC 2007 (converted to HHV)
Total Energy/ Unit	1.62	1.89	
	Cons	teel Evolution Valu	Ies*
Natural gas	0.28	0.33	Giavani et al. 2012
Coal	0.52	0.60	Giavani et al. 2012
Electricity	0.92	1.07	Giavani et al. 2012
Total Energy/ Unit	1.79*	2.08*	

\* Not referenced for EAF SOA, provided for additional information only.

Table A2 15 State

Table A3-16. State of the Art Energy Intensity Estimate for Direct Reduced Ironmaking					
Energy Source Energy Intensity Energy Intensity Reference (MMBtu/ton) (GJ/MT)					
Natural Gas	8.06	9.38	Energiron 2013		
Electricity	0.26	0.31	Energiron 2013		
Total Energy/ Unit	8.33	9.68			

Table A3-17. State of th	e Art Energy Intensity	y Estimate for Contin	uous Casting
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Electricity	0.03	0.03	Based on NRC 2007 (converted to onsite energy)
Natural Gas	0.03	0.03	Based on NRC 2007 (converted to HHV)
Total Energy/ Unit	0.05	0.06	
	Th	in Slab Casting*	
Fuel	0.04	0.05	LBNL 2008 Table 2.1.3
Electricity	0.13	0.15	LBNL 2008 Table 2.1.3
Total Energy/ Unit	0.17	0.20	

### Table A3-17. State of the Art Energy Intensity Estimate for Continuous Casting

\* Not referenced for Casting SOA Energy Intensity, provided for additional information only.

Table A3-18. State of the Art Energy Intensity Estimate for Cold Rolling Mill												
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference									
Natural Gas	0.73	0.81	Based on LBNL 2008 Section 2.1.6 (converted to HHV)									
Electricity	0.41	0.48	Based on LBNL 2008 Section 2.1.6 (converted to onsite energy)									
Steam	0.31	0.36	Based on LBNL 2008 Section 2.1.6									
Total Energy/ Unit	1.42	1.65										

Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference					
		Hot Strip Mill						
Natural Gas	1.19	1.38	Based on NRC 2007 (converted to HHV)					
Electricity	0.24	0.28	Based on NRC 2007 (converted to onsite energy)					
Steam	0.03	0.04	Based on NRC 2007					
Other	0.01	0.01	Based on NRC 2007					
Recovery (Credit)	-0.03	-0.04	Based on NRC 2007					
Total Energy/ Unit	1.43	1.66						
		Plate Mill						
Fuel	1.17	1.36	Based on NRC 2007 (converted to HHV)					
Electricity	0.25	0.29	Based on NRC 2007 (converted to onsit energy)					
Recovery (Credit)	-0.13	-0.15	Based on NRC 2007					
Total Energy/ Unit	1.29	1.50						
	Li	ght Section (Rod) Mil	11					
Natural Gas	1.52	1.77	Based on NRC 2007 (converted to HHV)					
Electricity	0.33	0.38	Based on NRC 2007 (converted to onsite energy)					
Total Energy/ Unit	1.85	2.15						
	Ме	dium Section (Bar) M	111					
Natural Gas	1.43	1.66	Based on NRC 2007 (converted to HHV)					
Electricity	0.25	0.29	Based on NRC 2007 (converted to onsite energy)					
Total Energy/ Unit	1.68	1.95						
	Heavy Sect	ion (Bloom and Struc	·					
Natural Gas	1.43	1.66	Based on NRC 2007 (converted to HHV)					
Electricity	0.31	0.36	Based on NRC 2007 (converted to onsite energy)					
Total Energy/ Unit	1.74	2.02						
	Average for H	lot Strip, Plate and Se	ection Mills *					
Fuel	1.35	1.57						
Electricity	0.28	0.32						
Steam	0.01	0.01						
Recovery, Other	-0.03	-0.04						
Total Energy/ Unit	1.60	1.86						

\* Average referenced for Hot Rolling SOA

### **THEORETICAL MINIMUM ENERGY INTENSITIES**

Table A3-20. Theoretical Minimum Energy Intensity Estimate for Sintering												
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference									
Fuel	1.03	1.20	Fruehan et al. 2000 Table 9									
Total Energy/ Unit	1.03	1.20										

Table A3-21. Theoretical Minimum Energy Intensity Estimate for Cokemaking												
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference									
Fuel	1.72	2.00	Fruehan et al. 2000 Table 9									
Total Energy/ Unit	1.72	2.00										

Table A3-22. Theoretical Minimum Energy Intensity Estimate for Blast Furnace Ironmaking												
Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference									
<i>Ideal Case<sup>1</sup></i> Fuel	8.43	9.81	Fruehan et al. 2000 Table 2									
<i>Typical Case<sup>2</sup></i> Fuel	8.96	10.42	Fruehan et al. 2000 Table 2									
Total Energy/ Unit	8.43	9.81										

<sup>1</sup> Ideal Case has no gangue or ash (this value was referenced for Blast Furnace TM)

<sup>2</sup> Typical Case includes gangue and ash.

# Table A3-23. Theoretical Minimum Energy Intensity Estimate for Basic Oxygen Furnace Steelmaking

Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
Exothermic (oxidation) (net value)	-0.64	-0.75	Fruehan et al. 2000 Text and Table 11 note
Total Energy/ Unit	-0.64	-0.75	

 Table A3-24. Theoretical Minimum Energy Intensity Estimate for Electric Arc Furnace

 Steelmaking

Energy Source	Energy Intensity (MMBtu/ton)	Energy Intensity (GJ/MT)	Reference
	Assuming 50/50	fuel/electric split	
Fuel	0.57	0.66	Fruehan et al. 2000 Table 5
Electricity	0.57	0.66	Fruehan et al. 2000 Table 5
Total Energy/ Unit	1.14	1.33	
	Assuming a	all electric <sup>a</sup>	
Electricity	1.14	1.33	Fruehan et al. 2000 Table 5
Total Energy/ Unit	1.14	1.33	

<sup>a</sup> Referenced for Blast Furnace TM

Table A3-25. Theoretical Minimum Energy Intensity to Produce Direct Reduced Iron at 1173K (900°C or 1620°F) Reduction Temperature for Selected Conditions

Ore	Product	Energy (MMBtu/ton)	Energy (GJ/MT)
Pure Fe <sub>2</sub> O <sub>3</sub> *	Fe	7.188	8.360
Fe <sub>2</sub> O <sub>3</sub> - 1.4% SiO <sub>2</sub>	Fe - 2% SiO <sub>2</sub>	7.206	8.380
Fe <sub>2</sub> O <sub>3</sub> - 1.4% SiO <sub>2</sub>	Fe - 2% SiO <sub>2</sub> - 8% FeO	6.793	7.900
Fe <sub>2</sub> O <sub>3</sub> - 1.5% SiO <sub>2</sub>	Fe - 2% SiO <sub>2</sub> - 8% FeO - 2% C	7.246	8.427
Fe <sub>2</sub> O <sub>3</sub> - 1.5% SiO <sub>2</sub>	Fe - 2% SiO <sub>2</sub> - 7.7% FeO - 6% C	8.110	9.432
Total Energy/ Unit		7.188	8.360

\* Referenced for DRI TM

Note: Full credit is assumed for off gas

Reference: Fruehan et al. 2000 Table 3

Table A3-2	6. Theoretical Min	imum Energy Inte	nsity to Roll Stee	I for Selected I	Products and Con	ditions							
Rolling Type	Rolling Temperature	Slab Temperature <sup>a</sup>	Reduction (mm)	(N	Energy IMBtu/ton), [GJ/M	T]							
Type	(K)	(K)	()	Heat	Deformation	Total							
		Flat Carbo	n Slab (25.4 cm c	or 10 inch)									
Hot	1473	298	254 to 2	0.709 [0.825]	0.021 [0.025]	0.731 [0.850]							
Hot	1473	1173	254 to 2	0.232 [0.270]	0.021 [0.025]	0.254 [0.295]							
Hot *	1473	1473	254 to 2		0.021 [0.025]	0.021 [0.025]							
Cold **	298	298	2 to 1		0.015 [0.017]	0.015 [0.017]							
Flat Carbon Slab (5.0 cm or 1.97 inch)													
Hot	1473	298	50 to 2	0.709 [0.825]	0.014 [0.016]	0.723 [0.841]							
Hot	1473	1173	50 to 2	0.232 [0.270]	0.014 [0.016]	0.246 [0.286]							
Hot	1473	1473	50 to 2		0.013 [0.015]	0.014 [0.016]							
Cold	298	298	2 to 1		0.015 [0.017]	0.015 [0.017]							
		Flat 18-8 Stainles	ss Steel Slab (25.	4 cm or 10 incl	n)								
Hot	1473	298	254 to 2	0.709 [0.825]	0.062 [0.072]	0.771 [0.897]							
Hot	1473	1173	254 to 2	0.233 [0.271]	0.062 [0.072]	0.294 [0.342]							
Hot	1473	1473	254 to 2		0.062 [0.072]	0.062 [0.072]							
Cold	298	298	2 to 1		0.044 [0.051]	0.044 [0.051]							
		Bar Carbo	n (10 cm billet to	2 cm bar)									
Hot	1473	298	10 sq to 2 sq	0.709 [0.825]	0.017 [0.020]	0.727 [0.845]							
Hot	1473	1473	10 sq to 2 sq		0.017 [0.020]	0.017 [0.020]							
Hot	1473	1173	10 sq to 2 sq	0.232 [0.270]	0.017 [0.020]	0.249 [0.290]							
Hot <sup>b</sup>	1473	1473	10 sq to 2 sq		0.009 [0.011]	0.009 [0.011]							
Total Energ	gy/ Unit (Hot Rollin	ng)				0.021 [0.025]							
Total Energ	gy/ Unit (Cold Roll	ling)				0.015 [0.017]							

Table A3-26. Theoretical Minimum Energy Intensity to Roll Steel for Selected Products and Conditions

\* Referenced for Hot Rolling TM, \*\* Referenced for Cold Rolling TM

<sup>a</sup> Slab temperature prior to rolling

<sup>b</sup> Billet split into 4 pieces prior to rolling

Reference: Fruehan et al. 2000 Table 7

				Fuel									Other	Other		Fuel, Elec Steam					
	Intensity	Energy	Feed- stock <sup>a</sup>	C	oal	Natura	al gas	Coke and Breeze		COG		BFG		Fue	l Oil	Elect	(utilites, credit)	Steam	Energy		
	Source	Intensity	Intensity	Intensity		LHV	HHV <sup>b</sup>	LHV	HHV <sup>b</sup>	LHV	HHV <sup>b</sup>	LHV	HHV <sup>c</sup>	LHV	HHV <sup>c</sup>	LHV	HHV <sup>b</sup>		or o any		(HHV)
Integrated Mil	ls																				
Agglomeration	Table A3-2	GJ/MT sinter				0.07	0.08	1.28	1.36							0.10			1.5		
Sintering		Percentage				55	%	88	%							7%			100%		
Pelletizing	Table A3-1	GJ/MT pellets				0.01	0.01	0.34	0.36	0.31	0.35					0.10			0.8		
relietizing		Percentage				19	%	44	%	429	%					12%			100%		
Cokemaking	Table A3-3	GJ/MT coke	33.6				0.39				3.24		0.04			0.12		0.66	4.4		
Cokemaking	Table A3-3	Percentage				9	%			739	%	19	%			3%		15%	100%		
Ironmaking	Table A3-4	GJ/MT pig iron			2.13		1.88		12.26		0.28		-3.4		0.16	0.27		0.05	13.6		
Blast Furnace	Table A3-4	Percentage		16	16%		14%		90%		2%		-25%		1%			0%	100%		
Steelmaking	g Table A3-5	GJ/MT steel				0.39	0.43					0.03	0.03			0.13		0.08	0.6		
BOF	Table A3-5	Percentage				64	%					4%				19%		12%	100%		
Mini Mills																					
Steelmaking	Table A3-6	GJ/MT steel					0.51									1.65			2.1		
EAF	Table A3-0	Percentage				24	%									76%			100%		
Integrated and	d Mini Mills	5																			
Casting	Table A3-8	GJ/MT steel				0.08	0.09									0.08	0.05		0.2		
Casting		Percentage				53	%									47%			100%		
Rolling Hot	Table A3-9	GJ/MT steel				2.275	2.52									0.44	0.04		3.0		
not		Percentage				85	%											0%	85%		
Cold	Table A3-10	GJ/MT steel				1.66	1.84							0.01		1.22		0.99	4.0		
		Percentage				45	%							09	%	30%		24%	100%		
<sup>a</sup> Feedstock coal	intensity bas	sed on [IPPC 20	13], 1285	kg dry (	coal/met	ric ton c	oke pro	duced.	Energy	alue of o	coal 22.	489 MN	Btu/shc	ort ton (E	EIA form	846).					

Table A3-27. Conversion from Lower Heating Value to Higher Heating Value for Current Typical Energy Intensity

<sup>c</sup> The Engineering Toolbox, www.engineeringtoolbox.com, Fuel Gases - Heating Values

					State	of the A	Art Ene	ergy Inte	ensity	by En	ergy Ty	ype an	d Proc	ess Ar	ea						
			E. J		Fuel														Other		Fuel, Elec Steam
	Intensity	Energy	Feed- stock <sup>a</sup>	Co	Coal		Natural gas		Coke and Breeze		COG		BFG		Gas	Fue	l Oil	Elect	(utilites, credit)	Steam	Energy Intensity
	Source	Intensity		LHV			HHV <sup>b</sup>	LHV F	HV p	LHV HHV °		LHV	HHV <sup>c</sup>	LHV	HHV <sup>d</sup>	LHV	HHV <sup>b</sup>				(HHV)
Integrated Mil	ls																				
Agglomeration	Table A3-11	GJ/MT sinter				0.06	0.07	1.23	1.30									0.10			1.47
Sintering	Table A3-TT	Percentage				5%	6	88%	6									7%			100%
Cokemaking	Table A3-12	GJ/MT coke	33.6			0.34	0.38			2.83	3.16	0.04	0.04					0.10		0.25	3.92
Cokemaking	Table AS-12	Percentage			10		10%				%	1	1%					3%		6%	100%
Ironmaking	Table A3-13	GJ/MT pig iron		2.55	2.69			10.83	11.49	0.23	0.26	-3.4	-3.40			1.32	1.41	0.09		0.40	12.94
Blast Furnace	Table AS-13	Percentage		21	%			89%		2	2% -2		5%			11%		1%		3%	100%
Steelmaking	Table A3-14	GJ/MT steel				0.36	0.40							-0.67	-0.67			0.09		-0.17	-0.35
BOF	Table AS-14	Percentage				NA								NA				NA		NA	NA
Mini Mills																					
Steelmaking	Table A3-15	GJ/MT steel				0.16	0.18											1.71			1.89
EAF		Percentage				9%	6											91%			100%
Integrated and	d Mini Mills																				
Casting	Table A3-17	GJ/MT steel				0.03	0.03											0.03			0.06
		Percentage				53	%											47%			100%
Rolling						4.443	4 57											0.00	0.04	0.01	1.86
Hot	Table A3-18	GJ/MT steel Percentage				1.416 83												0.32 17%	-0.04	0.01 0%	1.80
	Table 43-10	GJ/MT steel				0.73	0.81											0.48		0.36	
Cold		Percentage				49										0	%	29%		22%	1.05
<sup>a</sup> Feedstock coal	intonaity boo	, j	121 1205	ka day a	ool/mot					oluo of a	2001 22 /		Ptu/aba	rt top (E	1A form	-		2070		2270	10070
I COUSIOUR COAL	intensity bas		10], 1200	ry ury c	Juai/meti		re hior	uudeu. El	iergy v		Judi 22.4	+09 IVIIVI	Diu/5110			040).					

#### Table A3-28. Conversion from Lower Heating Value to Higher Heating Value for State of the Art Energy Intensity

<sup>b</sup> Lower heating value intensity converted to higher heating value using HHV/LHV ratio, GREET 1.8d.1, The Greenhouse Gases, Regulated Emissions, and Energy Use in the Transportation Model developed by ANL, 2010, Appendix A. Coke and Breeze LHV and HHV not available, Petroleum Coke ratio used in place. Fuel Oil assumed to be Residual Oil.

<sup>c</sup> The Engineering Toolbox, www.engineeringtoolbox.com, Fuel Gases - Heating Values

<sup>d</sup> Institute for Industrial Productivity, Industrial Efficiency Technology Database, BOF Heat and Gas Recovery: http://ietd.iipnetwork.org/content/bof-heat-and-gas-recovery, 8.8 MJ/Nm3. Assuming equivalent LHV and HHV.

## Appendix A4: Technologies Analyzed to Estimate Practical Minimum Energy Intensities with References

Table A4. Technologies Analyzed to Estimate Practical Minimum Energy Intensities							
Technology Name	Technology Description	Applicabi- lity (Product, process)	Source (See Reference list at end)	Reported Energy Savings (Literature- reported savings, Btu, %, etc.)	Explanation of Savings Baseline, or Reference (Adjustment, conversion, scale up of reported savings)	Calculated Product/ Process Savings (Savings compared to SOA or CT energy use. PM savings estimate.)	PM Energy Intensity (MMBtu/ ton) or % savings
Cokemaking							
Single- chamber- system coking reactors	Replace series of coking ovens with a single large volume oven	Cokemaking	Díez et al. 2002, IPPC 2001, EPA 2012, Nashan 2007, LBNL 2010	38% to 70%	Thermal efficiency improvement from 38% to 70%	CT thermal intensity for cokemaking = 5.02 MMBtu/ton . 38-70% savings claimed; assume lower (38%). 38% savings over CT = 3.83 x 0.38 = 1.46 MMBtu/ton	2.37 MMBtu/ ton
Coal Moisture Control	Drying of coal with waste heat gases	Cokemaking	APP 2010	0.3 GJ/tonne	Fuel savings of 0.3 GJ/tonne	EPA 2012 document estimates fuel savings of 0.268 MMBtu/ton (0.3 GJ/tonne). PM would be equal to 5.02268 = 4.75 MMBtu/ton	3.56 MMBtu/ ton
Blast Furnace	e Ironmaking						
Top Pressure Recovery Turbines	<ul> <li>Use hot high pressure gas from the furnace to power a turbine</li> <li>Dust removal from blast furnace gases using dry and wet methods.</li> </ul>	Ironmaking – Blast Furnace	APP 2010, EPA 2012, Inoue 1995, NEDO 2008, Stelco 1993	Additional 14- 36 kWh/ton metal produced	Turbine could produce additional 14-36 kWh/ton of hot metal (depending on available pressure)	EPA 2012 estimates 0.095 MMBtu/ton (0.11 GJ/tonne) electricity savings.	11.63 MMBtu/ ton

				Reported Energy		Calculated Product/	PM Energy
Technology Name	Technology Description	Applicabi- lity (Product, process)	Source (See Reference list at end)	Savings (Literature- reported savings, Btu, %, etc.)	Explanation of Savings Baseline, or Reference (Adjustment, conversion, scale up of reported savings)	Process Savings (Savings compared to SOA or CT energy use. PM savings estimate.)	Intensity (MMBtu/ ton) or % savings
Blast Furnace	e Ironmaking (continued)						
Top Pressure Recovery Turbines	<ul> <li>Use hot high pressure gas from the furnace to power a turbine</li> <li>Dust removal from blast furnace gases using dry and wet methods.</li> </ul>	Ironmaking – Blast Furnace	APP 2010, EPA 2012, Inoue 1995, NEDO 2008, Stelco 1993	Additional 14- 36 kWh/ton metal produced	Turbine could produce additional 14-36 kWh/ton of hot metal (depending on available pressure)	EPA 2012 estimates 0.095 MMBtu/ton (0.11 GJ/tonne) electricity savings.	11.63 MMBtu/ ton
Drum Chute and Segregation Slit Charging	<ul> <li>Use of a drum chute and segregation slit wire to control the particles dropping into the furnace</li> </ul>	Ironmaking – Blast Furnace	EPA 2012, NEDO 2008, LBNL 2010	0.7 MMBtu/ton	Can decrease coke use by 0.7 MMBtu/ton	Reported savings of 0.7 MMBtu/ton (0.8 GJ/tonne).	11.02 MMBtu/ ton
Heat recovery from blast furnace slag	<ul> <li>Capture of embedded heat in blast furnace slag through recovery as hot air or steam, chemical energy, or thermoelectric power</li> </ul>	Ironmaking – Blast Furnace	Barati et al. 2011, IPPC 2013, JISF 2012, LBNL 2010, POSCO 2010	0.35 GJ/tonne	Savings of approximately 0.35 GJ/tonne of pig iron	Reported savings of 0.3 MMBtu/ton (0.35 GJ/tonne).	11.42 MMBtu/ ton
BOF Steelma	king				1		
Recycling and reuse of basic oxygen furnace slag	<ul> <li>Separates BOF slag into three products allowing greater iron recovery and recycling</li> <li>Use of low-grade iron byproduct for acid mine neutralization</li> </ul>	BOF Steelmaking	DOE 2002a, IMP 2006, Energetics 2005	0.12 MMBtu/ton	0.12 MMBtu/ton (0.14 GJ/tonne) savings estimated	Estimated savings of 0.12 MMBtu/ton (0.14 GJ/tonne)	0.46 MMBtu/ ton

80 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

				Reported			
Technology Name	Technology Description	Applicabi- lity (Product, process)	Source (See Reference list at end)	Energy Savings (Literature- reported savings, Btu, %, etc.)	Explanation of Savings Baseline, or Reference (Adjustment, conversion, scale up of reported savings)	Calculated Product/ Process Savings (Savings compared to SOA or CT energy use. PM savings estimate.)	PM Energy Intensity (MMBtu/ ton) or % savings
EAF Steelma	king				^	^	
EPC System for Side Charging and Scrap Preheating	<ul> <li>Design to allow continuous charging of preheated scarp into the EAF. Separation of preheating and cold scrap charging.</li> <li>A design that allows movable scrap preheater to allow operation with or without use of scrap preheating</li> <li>Reduction in gas flows, totally sealed system and substantial reduction in dust and other emissions.</li> </ul>	EAF Steelmaking	KR Tec n.d., Rummler et al. n.d.	100 kWh/tonne	Preheating of scrap up to 700 °C reduces EAF energy consumption by up to 100 Kwh/tonne of molten steel.	Reported savings of up to 100 kWh/metric ton; but is similar to Consteel Evolution preheating. Assume 30 kWh/ton (0.1 MMBtu/ton - or 0.12 GJ/tonne).	1.76 MMBtu/ ton
Contiarc Furnace	Replace ladle metallurgy furnace with a continuous series of vessels	EAF Steelmaking	AEHOF 2013, EPA 2012, IPPC 2013	200 kWh/ton	Reduced energy losses (200 kWh/ton) over conventional furnace	EPA 2012 document estimates 0.62 MMBtu/ton (0.72 GJ/tonne) electricity savings.	1.24 MMBtu/ ton
Casting					·	·	
Continuous Casting for EAF	Replace ladle metallurgy furnace with a continuous series of vessels	Casting	Peaslee et al. 2006, DOE 2005	10%	Anticipated 10% decrease in energy consumption	10% savings estimate (or 0.005 MMBtu/ton - 0.006 GJ/tonne)	0.19 MMBtu/ ton
Tundish Heating Technologies (Cold Tundish)	<ul> <li>Using a cold tundish</li> <li>Heating a tundish inductively and not by combustion</li> </ul>	Casting	Beraldo et al. 2003, EPA 2012, LBNL 2010	78% (natural gas)	78% decrease in natural gas usage	While impact may vary by plant, EPA 2012 document estimates 0.017 MMBtu/ton (0.02 GJ/tonne) savings.	0.17 MMBtu/ ton

Technology Name	Technology Description	Applicabi- lity (Product, process)	Source (See Reference list at end)	Reported Energy Savings (Literature- reported savings, Btu, %, etc.)	Explanation of Savings Baseline, or Reference (Adjustment, conversion, scale up of reported savings)	Calculated Product/ Process Savings (Savings compared to SOA or CT energy use. PM savings estimate.)	PM Energy Intensity (MMBtu/ ton) or % savings
Rolling							
Endless Rolling	New development in thin slab casting and direct rolling	Rolling	Arvedi et al. 2008, EPA 2012	40%	Anticipated 40% lower energy than a traditional rolling mill	40% savings estimate	1.55 – 2.09 MMBtu/ ton
Next- Generation System for Scale-free Steel Reheating	<ul> <li>Use of preheated or oxygen- enriched air to control flue gas</li> <li>Improves the quality and yield of steel while increasing energy and production efficiency</li> </ul>	Rolling	Thekdi 2010, DOE 2010	0.2 GJ/tonne	<ul> <li>22-32% of current energy used for reheating</li> <li>0.2 GJ/tonne during reheating</li> </ul>	Estimated savings of 0.17 MMBtu/ton (0.2 GJ/tonne).	2.41 - 3.31 MMBtu/ ton
High temperature insulation materials	Innovative insulating materials will limit their consumption in a furnace	Rolling	BMWi 2008, EPA 2012	30-35% possible	<ul> <li>Energy savings of 30-35% are possible</li> <li>Likely savings of 2-5% on furnaces</li> </ul>	EPA 2012 document estimates fuel savings of 0.14 MMBtu/ton (0.16 GJ/tonne).	2.44 – 3.34 MMBtu/ ton
Crosscutting	Technologies		1		1		
New High- Temperature, Low-Cost Ceramic Media for Natural Gas Combustion Burners	Combining four different technologies into a single radiant burner package that functions as both a burner and a catalyst support.	Could potentially apply when electric or natural gas radiant heaters used in process heating.	DOE 2011	25% reduction in energy for process heat	Potential to reduce energy consumption by 25% for process heat.	From MECS data, 626 TBtu of direct end use for process heating. This equates to 64% of direct end use. 25% savings of 83% energy use results in 16% average savings. Practical minimum specific energy savings of 16% over CT applied to all process areas.	16% savings over CT for all process areas

82 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing

Table A4. Technologies Analyzed to Estimate Practical Minimum Energy Intensities							
Technology Name	Technology Description	Applicabi- lity (Product, process)	Source (See Reference list at end)	Reported Energy Savings (Literature- reported savings, Btu, %, etc.)	Explanation of Savings Baseline, or Reference (Adjustment, conversion, scale up of reported savings)	Calculated Product/ Process Savings (Savings compared to SOA or CT energy use. PM savings estimate.)	PM Energy Intensity (MMBtu/ ton) or % savings
Crosscutting	Technologies (continued)						
Advanced Energy and Water Recovery Technology from Low- Grade Waste Heat	Recovery of high purity water and energy from low grade heat, high moisture waste streams using nanoporous membranes. Will prove concept in laboratory and evaluate in "two different types of industrial environments.	Applies to any process step that produces sufficient low-grade waste heat to make the process viable	DOE 2011c; GTI 2011	20-30% greater energy efficiency in recovery from low grade waste heat.	The amount of energy savings would depend on the amount of waste heat could be recovered. Using the nanoporous membrane technology could increase heat recovery by 20- 30% it would appear.	There will be an estimated 5.7 TBtu/year energy savings for the iron and steel industry with wet scrubbers. Compared to the overall CT energy consumption of 981 TBtu/year, this represents a 1% energy savings. Practical minimum energy savings of 1% over CT is applied to all processes.	1% savings over CT for all process areas
Control Systems for Energy- efficient Recycling of Steel Residues	By utilizing computer-aided control of the process conditions, changes can be made in the prevailing conditions to reduce recycling energy consumption. In practical trials this software helped reduce energy consumption by 10%. The modular design of this software also enables changes to the processing conditions to reflect the desired product quality.	Crosscutting	BMWi 2008	10% reduction in energy consumption	Opportunity reduced to 2%.	2% reduction in energy consumption assumed	2% savings over CT for all process areas

The four bandwidth measures are current typical (CT), state of the art (SOA), practical minimum (PM), and thermodynamic minimum (TM).

### **REFERENCES FOR TABLE A4**

AEHOF 2013	"ACIPCO's 65-MW Contiarc Furnace." Alabama Engineering Hall of Fame (AEHOF). 2013. <u>http://aehof.eng.ua.edu/members/acipcos-65-megawatt- contiarc-furnace/</u>
AISI 2009	<i>TRP</i> 9956 – A Technical Feasibility Study of Steelmaking by Molten Oxide Electrolysis. Sadoway, D.R. & Ceder, G. for the American Iron and Steel Institute (AISI) and U.S. Department of Energy (DOE). 2009. www.steeltrp.com/finalreports/finalreports/9956NonPropFinalReport.pdf
AISI 2010	<i>Technology Roadmap Research Program for the Steel Industry</i> . American Iron and Steel Institute (AISI). 2010. <u>www.steel.org/en/sitecore/content/Global/Document%20Types/News/2011/~/</u> <u>media/Files/AISI/Making%20Steel/TechReportResearchProgramFINAL.ashx</u>
Allendorf et al. 2003	Optical Sensors for Post Combustion Control in Electric Arc Furnace Steelmaking. Allendorf, S.W., Ottesen, D.K., Green, R.W., & Hardesty, D.R. 2003. www.osti.gov/bridge/servlets/purl/840948-R3Plg4/native/840948.pdf
APP 2010	<i>The State–of-the-Art Clean Technologies (SOACT) for Steelmaking</i> <i>Handbook, 2nd Edition.</i> Asia Pacific Partnership on Clean Development and Climate (APP). 2010. <u>www.asiapacificpartnership.org/pdf/Projects/Steel/SOACT-Handbook-2nd- Edition.pdf</u>
Arvedi et al. 2008	"The Arvedi endless strip production line (ESP) from liquid steel to hot-rolled coil in seven minutes." Arvedi, G., Mazzolari, F., Bianchi, A., Holleis, G., Siegl, J, & Angerbauer, A. <i>La Revue de Métallurgie-CIT</i> (July/August): 398–407. 2008.
Asmet 2011	"New Pig Iron Process 'Tecnored Technology' Tested by Vale." Asmet. 2011. <u>www.asmet.com/news/new-pig-iron-process-tecnored-technology-tested-by-vale</u>
Barati et al. 2011	"Energy recovery from high temperature slags." Barati, M., S. Esfahani, & T. A. Utigard. <i>Energy</i> 36(9): 5440–49. 2011.
BCS 2005	Advanced Melting Technologies: Energy Saving Concepts and Opportunities for the Metal Casting Industry. BCS, Incorporated. 2005. www.energy.gov/sites/prod/files/2013/11/f4/advancedmeltingtechnologies.pd <u>f</u>
Becerra & Duarte 2007	HYL ENERGIRON - The Most Economical Way to High Quality DRI. Becerra, J. & Duarte, P. Arab Iron & Steel Union Conference. 2007.

Beraldo et al. 2003	"Implementation of Cold Tundish at CTS Continuous Casting Process." Beraldo, A.K., Moreira, D.B., Duarte, J.C., Matos, M.F., & Nascimento, Z.D. La Revue de Métallurgie-CIT. October, 955-960. 2003.
Birat et al. 2003	"CO <sub>2</sub> Mitigation Technologies in the Steel Industry: a Benchmarking Study Based on Process Calculations." Birat J. P., Hanrot, F., and Danloy, G. Stahl und Eisen, 123(9): 69–72. 2003.
Birat et al. 2008	<i>ULCOS Program: A Progress Report in the Spring of 2008.</i> Birat, JP., et al. 3 <sup>rd</sup> International Conference on Process Development in Iron and Steelmaking, Sweden, June 8-11. 2008. <u>www.ulcos.org/en/docs/Ref09%20-%20ULCOS_Scanmet%20III.pdf</u>
BMWi 2008	Climate Protection and Energy Efficiency: Research, Development and Demonstration of Modern Energy Technologies. German Federal Ministry of Economics and Technology (BMWi). 2008. www.bmwi.de/English/Redaktion/Pdf/climate-protection-energy- efficiency,property=pdf,bereich=bmwi,sprache=en,rwb=true.pdf
Boyd & Sikka 2006	Aluminum Bronze Alloys to Improve the System Life of Basic Oxygen and Electric Arc Furnace Hoods, Roofs and Side Vents. Boyd Jr., L.C. & V.K. Sikka. Energy Industries of Ohio: Independence, Ohio, July 1. 2006. www.osti.gov/bridge/servlets/purl/896794-fYjaYy/896794.pdf
Castrip 2012	"The Castrip® Process." Castrip. 2012. www.castrip.com/Process/process.html
Cotchen & Strecho 2003	"Continuous Horizontal Sidewall (CHS) Scrap Charging" Cotchen, J.K. & M.J. Strecho. <i>ISSTech 2003 Conference Proceedings</i> . 2003. <u>http://digital.library.aist.org/pages/PR-007-069.htm</u>
Danieli Corus. n.d.	"The Tecnored Process A New Route For Producing Low Cost Liquid Iron." Danieli Corus. n.d. <u>ftp://ftp.cefetes.br/Cursos/EngenhariaMetalurgica/Marcelolucas/Disciplinas/R</u> <u>educao/Material%20Tecnored/Brochure.pdf</u>
de Beer et al. 1998	<i>Future Technologies for Energy-Efficient Iron and Steel Making.</i> de Beer, J., Worrell, E., & Blok, K. <i>Annu. Rev. Energy Environ.</i> 23:123-205. 1998. <u>www.annualreviews.org/doi/abs/10.1146/annurev.energy.23.1.123</u>
Diez et al.	"Coal for Metallurgical Coke Production: Predictions of Coke Quality and
2002	Future Requirements for Cokemaking." Diez, M. A., R. Alvarez, & C. Barriocanal. <i>International Journal of Coal Geology</i> 50(1–4): 389–412. 2002.

DOE 2002b	Novel Direct Steelmaking by Combining Microwave, Electric Arc, and Exothermal Heating Technologies. U.S. Department of Energy (DOE) Office of Industrial Technology (OIT). 2002. www.osti.gov/scitech/servlets/purl/840931
DOE 2002c	Nitrogen Control in Electric Arc Furnace Steelmaking by Direct Reduced Iron Fines Injection. U.S. Department of Energy (DOE) Office of Industrial Technology (OIT). 2002. www.energy.gov/sites/prod/files/2014/06/f16/nitrogen_control.pdf
DOE 2004	<i>Ironmaking: Quality and Supply Critical to Steel Industry.</i> U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2004. <a href="https://www.energy.gov/sites/prod/files/2014/05/f16/mesabi_success.pdf">www.energy.gov/sites/prod/files/2014/05/f16/mesabi_success.pdf</a>
DOE 2005	Process to Continuously Melt, Refine, and Cast High Quality Steel. U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2005. www.energy.gov/sites/prod/files/2014/05/f16/dev_of_process.pdf
DOE 2007a	Improving the System Life of Basic Oxygen and Electric Arc Furnace Hoods, Roofs and Side Vents. U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2007. www.energy.gov/sites/prod/files/2014/05/f16/improving_system_life.pdf
DOE 2007b	Next Generation Metallic Iron Nodule Technology in Electric Furnace Steelmaking. U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2007. www.energy.gov/sites/prod/files/2014/06/f16/iron_nodule.pdf
DOE 2010	<i>IMPACTS: Industrial Technologies Program: Summary of Program Results for CY 2009.</i> U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2010. www1.eere.energy.gov/manufacturing/about/pdfs/impacts2009_full_report.pd <u>f</u>
DOE 2011a	Paired Straight Hearth Furnace. U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2011. www.energy.gov/sites/prod/files/2013/11/f4/paired_straight_hearth_furnace.pdf
DOE 2011b	<i>Thermochemical Recuperation for High Temperature Furnaces</i> . U.S. Department of Energy (DOE) Industrial Technologies Program (ITP). 2011. www.energy.gov/sites/prod/files/2014/05/f16/thermochemical_recuperation.p df
DOE 2011c	Grand Challenge Portfolio: Driving Innovations in Industrial Energy Efficiency. U.S. Department of Energy (DOE), Industrial Technologies Program (ITP). 2011. www1.eere.energy.gov/manufacturing/pdfs/grand_challenges_portfolio.pdf

EIA 2013	Model Documentation Report: Industrial Demand Module of the National Energy Modeling System. U.S. Energy Information Administration (EIA). 2013. www.eia.gov/forecasts/aeo/nems/documentation/industrial/pdf/m064%28201 3%29.pdf
Energetics 2005	<i>Steel Industry Marginal Opportunity Study</i> . Energetics, Inc. for U.S. Department of Energy (DOE), Industrial Technologies Program (ITP). 2005. <u>www.energy.gov/sites/prod/files/2013/11/f4/steelmarginalopportunity.pdf</u>
EPA 2012	Available and Emerging Technologies for Reducing Greenhouse Gas Emissions from the Iron and Steel Industry. U.S. Environmental Protection Agency (EPA). 2012. <u>www.epa.gov/nsr/ghgdocs/ironsteel.pdf</u>
Fosnacht et al. 2010	<i>Next Generation Metallic Iron Nodule Technology in Electric Arc</i> <i>Steelmaking – Phase II.</i> Fosnacht, D. R., et al. Natural Resources Research Institute. 2010. <u>www.osti.gov/bridge/servlets/purl/1000412-</u> <u>Jcpwcd/1000412.pdf</u>
GTI 2011	Final Technical Report: Advanced Energy and Water Recovery Technology from Low Grade Waste Heat. Gas Technology Institute (GTI). 2011. www.osti.gov/bridge/servlets/purl/1031483/1031483.pdf
Hacala & Michon 2009	"Innovative Industrial Plasma Torch for Converting Biomass Into High Purity Syngas." Hacala, A. & U. Michon. International Plasma Chemistry Society Conference 19, Bochum, Germany. 2009. <u>www.ispc- conference.org/ispcproc/papers/39.pdf</u>
Harada et al. 2005	"The Development of New Iron Making Processes." Harada, T., Tanaka, H. & Uemura, H. <i>Kobelco Technology Review</i> 26 (December). 2005.
Huang & Hwang 2005	Novel Direct Steelmaking by Combining Microwave, Electric Arc and Exothermal Heating Technologies. Huang, X. & J. Y. Hwang. 2005. www.osti.gov/bridge/servlets/purl/840931-0KppD1/native/840931.pdf
IETD 2013	"Iron and Steel." Industrial Efficiency Technology Database (IETD). 2013. www.ietd.iipnetwork.org/content/iron-and-steel
IMP 2006	Verification of Steelmaking Slag Iron Content, Final Technical Progress Report. DE-FC36-01ID14046. Institute of Materials Processing (IMP), Michigan Technological University. 2006. www.ntis.gov/assets/pdf/st-on-cd/DE2006892748.pdf
Inoue 1995	"The Steel Industry in Japan: Progress in Continuous Casting." In Energy Efficiency Utilizing High Technology: As Assessment of Energy Use in Industry and Buildings, Appendix A: Case Studies. Inoue, K. 1995.

Irons et al. 2004	Nitrogen Control in Electric Arc Furnace Steelmaking by DRI Fines Injection. Irons, G. A., Anghelina, D., & Brooks, G.A. for the American Iron and Steel Institute. 2004. <u>www.osti.gov/bridge/servlets/purl/840951-</u> gXP6N9/native/840951.pdf
IPPC 2001	<i>Reference Document on Best Available Techniques for the Production of Iron and Steel.</i> European Union. Integrated Pollution Prevention and Control Bureau (IPPC). 2001.
IPPC 2013	Best Available Techniques (BAT) Reference Document for Iron and Steel Production. European Commission. Integrated Pollution Prevention and Control (IPPC). 2013. http://eippcb.jrc.ec.europa.eu/reference/BREF/IS_Adopted_03_2012.pdf
JCOAL 2007	<i>Clean Coal Technologies in Japan</i> . Japan Coal Energy Center (JCOAL). 2007. <u>www.jcoal.or.jp/cctinjapan_en/cctinjapan_en.html</u>
Jernkontoret 2013	"IronArc revolutionises iron production – carbon dioxide emissions halved." Press release. Jernkontoret. 2013. www.jernkontoret.se/om_oss/pressmeddelanden/pdf_2013/press130206_iron arc_eng.pdf
JFE 2004	<i>Ecologically-Friendly and Economical Arc Furnace "ECOARC"</i> . JFE Engineering. 2004. <u>www.jfe-steel.co.jp/en/research/report/003/pdf/003-16.pdf</u>
JISF 2012	"Technologies to Reduce CO <sub>2</sub> Emissions." Japan Iron and Steel Federation (JISF). 2012. <u>www.jisf.or.jp/course50/tecnology01/index_en.html</u>
Júniora et al. 2005	"Tecnored Process - High Potential in Using Different Kinds of Solid Fuels." Júniora, J.H.N., Contruccib, M.A., & D'Abreuc, J.C. <i>Materials Research</i> 8(4): 447–51. 2005. <u>www.scielo.br/scielo.php?pid=S1516-</u> <u>14392005000400016&amp;script=sci_arttext</u>
KR Tec n.d.	Environmental Preheating and Continuous Charging System (EP). KR Tec and CVS Technology. n.d.
Krause 2008	<i>Energy Efficiency in Manufacturing: The Role of ICT</i> . Krause, I.F-L. for ICT and Energy Efficiency Consultation Group on Smart Manufacturing. 2008. <u>ftp://ftp.cordis.europa.eu/pub/fp7/ict/docs/20080703-smart-manufacturing-report-final_en.pdf</u>
LBNL 2010	Energy Efficiency Improvement and Cost Saving Opportunities for the Iron and Steel Industry: An ENERGY STAR Guide for Energy and Plant Managers. Neelis, M., Worrell, E., & Masanet, E. for Lawrence Berkeley National Laboratory (LBNL). 2010. <u>http://eetd.lbl.gov/sites/all/files/lbnl-</u> <u>4779e.pdf</u>

Lockwood Greene 2000	<i>Iron Making Process Alternatives Study</i> . Lockwood Greene for the U.S. Department of Energy. 2000. www.energy.gov/sites/prod/files/2013/11/f4/ironmaking_process.pdf
Lu 2008	"The Search for a New Ironmaking Process Towards Sustainable Development." Lu, W.K. GCEP "Industry" Workshop, Stanford University, April 15-16. 2008. <u>http://gcep.stanford.edu/pdfs/2RK4ZjKBF2f71uM4uriP9g/Lu_New_Sustaina ble_Ironmaking.pdf</u>
Midrex 2013	"Reducing Gas-Coal (MXCOL)." Midrex. 2013. www.midrex.com/handler.cfm/ cat_id/171/section/global
Nashan 2007	Strategies to Increase the Technical and Economic Efficiency of the Blast Furnace Ironmaking Process by Using New Advanced Cokemaking Technologies. Nashan, G. 2007. <u>www.coke-oven-</u> <u>managers.org/PDFs/2007/2007%20-%20Nashan.pdf</u>
NEDO 2008	Global Warming Countermeasures: Japanese Technologies for Energy Savings/GHG Emissions Reduction, 2008 Revised Edition. New Energy and Industrial Technology Development Organization (NEDO). 2008. www.nedo.go.jp/content/100107259.pdf
NEDO 2010	<i>Technology Development for Environmentally Harmonized Steelmaking</i> <i>Processes.</i> New Energy and Industrial Technology Development Organization (NEDO). Workshop on Delivering Green Growth, Seoul, South Korea, March 4. 2010. <u>www.oecd.org/sti/industryandglobalisation/45010258.pdf</u>
NEDO 2011	Annual Report 2011. New Energy and Industrial Technology Development Organization (NEDO). 2011. <u>www.nedo.go.jp/content/100185546.pdf</u> .
Peaslee et al. 2006	Development of a Process to Continuously Melt, Refine and Cast High Quality Steel. Final Technical Report. Peaslee K. D., Richards, V.L., Robertson, D.G.C., & Smith. J.D. for the U.S. Department of Energy. 2006. www.energy.gov/sites/prod/files/2014/05/f16/dev_of_process.pdf
Pokhvisnev et al. 2002	"Romelt Process- Promising Technology for Ironmaking." Pokhvisnev, Y., Romenets, V., Valavin, V., Zaytsev, A., & Vandariev, S. <i>Metal.</i> 2002. <u>http://konsys-t.tanger.cz/files/proceedings/metal_02/papers/100.pdf</u>
POSCO 2008	"The FINEX Innovative Ironmaking Process." Pohang Iron & Steel Co, Ltd. (POSCO). The Asia-Pacific Partnership on Clean Development and Climate, 4th Steel Task Force Workshop. 2008. www.asiapacificpartnership.org/pdf/Steel/5th_meeting/1-2(POSCO).pdf
POSCO 2009	2009 Carbon Report. Pohang Iron & Steel Co, Ltd. Pohang Iron & Steel Co, Ltd. (POSCO). 2009.

POSCO 2010	<i>Carbon Report 2010.</i> Pohang Iron & Steel Co, Ltd. Pohang Iron & Steel Co, Ltd. (POSCO). 2010.
Rummler et al. n.d.	"New Generation in preheating technology for electric steel making - higher productivity with reduced power." Rummler, K., Tunaboylu, A. & Ertas, D. Sundheim, Germany: KR Tech GMBH. n.d.
Siemens VAI 2013	"Finex Technology." Siemens VAI. 2013. www.industry.siemens.com/verticals/metals- industry/en/metals/ironmaking/finex/pages/home.aspx
Sosinsky et al. 2008	"The CASTRIP® process – recent developments at Nucor Steel's commercial strip casting plant." Sosinsky, D., Campbell, P., Mahapatra, R., Blejde, W., & Fischer, F. <i>Metallurgist</i> 52(11–12): 691–99. 2008.
Stelco 1993	Present and Future Use of Energy in the Canadian Steel Industry. Ottawa, Canada: CANMET. Stelco. 1993.
Tanaka 2007	"New Iron making Processes for CO <sub>2</sub> Reduction." Tanaka, H. Asia-Pacific Partnership on Clean Development and Climate 4th Steel Workshop, Wollongong Australia, October 25. 2007.
Tenova HYL 2008	"Improved Technologies for Cost Effective Steelmaking," Tenova HYL. Metal Bulletin/MJunction, Indian Steel 2008 Conference, Mumbai, India, February. 2008.
Thekdi 2010	Development of Next Generation Heating System For Scale Free Steel Reheating-Final Technical Report. Thekdi, A. for the U.S Department of Energy. 2010. <u>www.osti.gov/scitech/servlets/purl/1004059</u>
ULCOS 2012	"ULCOWIN: Alkaline Electrolysis." Ultra-Low CO <sub>2</sub> Steelmaking (ULCOS). 2012. <u>www.ulcos.org/en/research/electrolysis.php</u>
van Boggelen 2012	"HIsarna, a Low CO <sub>2</sub> Ironmaking Process." van Boggelen, J. Presentation at Cleveland Institution of Engineers, Middlesbrough, UK, January 10. 2012. www.cieng.org/events/lectures 2011 - 2012 session/ 2012-01-10 hisarna/2012 01 10 for publication v3.pdf
Wolfe 2009	"A continuous carbonite process for making foundry size formed coke." Wolfe, R. <i>Ductile Iron News</i> Issue 2. 2009.
Zuliani et al. 2010	"Opportunities for increasing productivity, lowering operating costs and reducing greenhouse gas emissions in EAF and BOF steelmaking." Zuliani, D. J., Scipolo, V. & Born, C. <i>Millennium Steel India 2010</i> . 2010. <u>http://millennium-steel.com/wp-</u> content/uploads/articles/pdf/2010%20India/pp35-42%20MSI10.pdf

# Appendix A5: Practical Minimum Technology Weighting Factors

### **METHODOLOGY TO DETERMINE WEIGHTING FACTORS**

In this section the practical minimum technology weighting factors methodology is explained. The application of this methodology is presented in Table A4.

Six Weighting Factors, A through F, are considered for each technology and scored as shown (High (H) = 3, Medium (M) = 2, Low (L) = 1, Not Available (NA) = 0). The factors are also scaled according to DOE Importance Level, e.g., an importance level of 2 carries twice the weight of an importance level of 1. For the iron and steel bandwidth, factors A-F each carried a DOE Importance Level of 1.

The DOE Importance Level is multiplied by the score for each factor and divided by the total possible score to determine overall weighting of technology. The NA score of 0 is excluded from overall weighting.

Factor A - Technology Readiness

- High = Technology Readiness Level (TRL) 7-9
- Medium = TRL 4-6
- Low = TRL 1-3

Factor B - Market Impact

- High = widely applicable to all establishments
- Medium = applicable to many establishments
- Low = applicable to select few establishments or unique process

Factor C - Relative Cost and Savings Payback

- High = implementation cost >90% of reference technology, or payback > 10 years
- Medium = cost <90% and >40% of reference technology, payback <10 years
- Low = cost <40% of reference, payback < 2 years

Note: the score is reversed such that H = 1 and L = 3

Factor D – Technical Risk

- High = high likelihood of technology success and deployment, minimal risk factors
- Medium = insufficient evidence of technology success, some risk factors
- Low = low likelihood of success, multiple and significant risk factors

Note: the score is reversed such that H = 1 and L = 3

Factor E – Productivity/Product Quality Gain

- High = significant gain in productivity, either quantity or quality of product produced
- Medium = moderate gain in productivity
- Low = no gain in productivity

Factor F - Environmental Benefits

- High = multiple and significant environmental benefits,
- Medium = some environmental benefits,
- Low = little or no environmental benefit

Importance		1	1 1		1			1		1		1	
Level		•											
Technology Name	A – Technology Readiness		B- Market Impact		C- Relative Cost and Savings Payback		eighting Factors D- Technical Risk		E – Productivity/ Product Quality Gain		F- Environmental Benefits		Overall Importance
	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	Rating
Cokemaking													
Single- chamber- system Coking Reactors	Μ	Engineering Judgment	н	Engineering Judgment	L	Engineering Judgment	Н	Engineering Judgment	Н	Engineering Judgment	L	Engineering Judgment	72%
Coal Moisture Control	М	Engineering Judgment	Н	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	Н	Engineering Judgment	М	Engineering Judgment	89%
Blast Furnace I	ronmaki	ing											
Top Pressure Recovery Turbines	Н	Engineering Judgment	М	Engineering Judgment	Н	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	н	Engineering Judgment	72%
Drum Chute and Segregation Slit Charging	Н	Engineering Judgment	М	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	Μ	Engineering Judgment	L	Engineering Judgment	78%
Heat Recovery From Blast Furnace Slag	М	Engineering Judgment	М	Engineering Judgment	Μ	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	М	Engineering Judgment	67%
BOF Steelmaki	ng												
Recycling and Reuse of Basic Oxygen Furnace Slag	н	Engineering Judgment	н	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	L	Engineering Judgment	Н	Engineering Judgment	78%

Table A5. Pract	ical Min	imum Technol	ogies A	nalysis with W	eighting	Factors							
Importance Level		1 1		1		1		1		1			
Technology Name	Technology Weighting Factors												
	A – Technology Readiness		B- Market Impact		C- Relative Cost and Savings Payback		D- Technical Risk		E – Productivity/ Product Quality Gain		F- Environmental Benefits		Overall Importance
	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	Rating
EAF Steelmaking													
EPC System for Side Charging and Scrap Preheating	Н	Engineering Judgment	М	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	83%
Contiarc Furnace	Н	Engineering Judgment	н	Engineering Judgment	н	Engineering Judgment	L	Engineering Judgment	н	Engineering Judgment	М	Engineering Judgment	83%
Casting													
Continuous Casting for EAF	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	н	Engineering Judgment	L	Engineering Judgment	67%
Tundish Heating Technologies (Cold Tundish)	Н	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	Н	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	56%
Rolling													
Endless Rolling	Н	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	н	Engineering Judgment	н	Engineering Judgment	80%
Next- Generation System for Scale-free Steel Reheating	Μ	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	Н	Engineering Judgment	72%

Importance Level		1 1		1		1		1		1			
Technology Name	Technology Weighting Factors												
	A – Technology Readiness		B- Market Impact		C- Relative Cost and Savings Payback		D- Technical Risk		E – Productivity/ Product Quality Gain		F- Environmental Benefits		Overall Importance
	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	H, M, or L	Explanation	Rating
High Temperature Insulation Materials	М	Engineering Judgment	н	Engineering Judgment	н	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	L	Engineering Judgment	61%
Crosscutting T	echnolo	gies											
New High- Temperature, Low-Cost Ceramic Media for Natural Gas Combustion Burners	н	Engineering Judgment - TRL 7	н	Wide ranging applications	М	Moderate capital investment	М	Moderate process change	М	Better heating	Н	Large energy savings	83%
Advanced Energy and Water Recovery Technology from Low- Grade Waste Heat	М	Engineering Judgment - TRL 4	Н	Wide ranging applications	Н	Major capital investment	Н	Large process change	L	Engineering Judgment	Н	Waste water recovery	61%
Control Systems for Energy- efficient Recycling of Steel Residues	н	Engineering Judgment - TRL 7	М	Engineering Judgment	М	Engineering Judgment	М	Engineering Judgment	н	Engineering Judgment	Н	Engineering Judgment	83%

Appendix A4 provides the methodology used to identify the weighting factors and the definitions for the abbreviations.

96 Bandwidth Study on Energy Use and Potential Energy Saving Opportunities in U.S. Iron and Steel Manufacturing



# U.S. DEPARTMENT OF

Energy Efficiency & Renewable Energy For more information, visit: manufacturing.energy.gov

DOE/EE-1231 • June 2015